# BANK SUPERVISION AND NON-PERFORMING LOAN CLEANSING

# 2024

BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2428

Soner Baskaya, José E. Gutiérrez, José María Serena and Serafeim Tsoukas

BANK SUPERVISION AND NON-PERFORMING LOAN CLEANSING

## BANK SUPERVISION AND NON-PERFORMING

## Soner Baskaya (\*\*)

UNIVERSITY OF GLASGOW

## José E. Gutiérrez (\*\*\*)

BANCO DE ESPAÑA

## José María Serena (\*\*\*\*)

BANCO DE ESPAÑA

## Serafeim Tsoukas (\*\*\*\*\*)

UNIVERSITY OF GLASGOW

(\*) This paper has been circulated previously under the title "Unloading NPLs, unlocking credit? Evidence from the CB provisioning guidelines." We would like to thank Korav Alper, Selim Elekdag, Co-Pierre Goerg, Gabriel Jiménez, David Martinez-Miera, Charles Nolan, Carlos Pérez Montes, José-Luis Peydró, our discussants Brunella Bruno, Angela Gallo, Giang Phung, Carlos Ramírez, and Barry Williams, and seminar participants at the Banco de España, the 2023 International Conference in Finance, Banking and Accounting in Montpellier, the 5th CRBF Conference in Contemporary Issues at Banking at the University of St. Andrews, the University of Strathclyde, the 2023 World Finance Banking Symposium, the 2024 Annual ECB Banking Supervision Research Conference, the 2024 Financial Management Association, and the 31st Finance Forum of the Spanish Finance Association for their comments. We also thank Shuren Shi for excellent research assistance. The first and fourth authors gratefully acknowledge support from the Banco de España and their generous hospitality during the period in which this paper was prepared. This article is the exclusive responsibility of its authors and does not necessarily reflect the official views of the Banco de España or the Eurosystem or their staff.

(\*\*) e-mail: soner.baskaya@glasgow.ac.uk (\*\*\*) e-mail: josee.gutierrez@bde.es

(\*\*\*\*) e-mail: josemaria.serena@bde.es

(\*\*\*\*\*) e-mail: serafeim.tsoukas@glasgow.ac.uk

Documentos de Trabajo. N.º 2428 September 2024

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2024

ISSN: 1579-8666 (on line)

#### Abstract

This paper studies whether supervisory actions, namely provisioning guidelines on nonperforming loans (NPLs), affect banks' NPL cleansing and lending behaviour, as well as the real economy. Using the supervisory intervention announced by the European Central Bank in the first quarter of 2018 as a quasi-natural experiment, we show that banks disposed of old NPLs at a higher rate after the policy shift. Banks that were more heavily exposed to the policy tightened their lending standards, especially for risky firms. Furthermore, banks with stronger fundamentals were more keen on disposing NPLs and less restrained on lending. We also find that firms borrowing from banks affected by the supervisory actions experienced a decline in the growth rates of their total assets, investment, employment and sales. Our results highlight the importance of supervisory actions on NPL management, and potential beneficial effects on credit allocation.

**Keywords:** non-performing loans, loan loss provisioning rules, NPL resolution, credit supply, firm outcomes.

JEL classification: E51, E58, G13, G21.

#### Resumen

Este documento estudia el impacto de las directrices supervisoras para los préstamos dudosos sobre su ritmo de provisión y el comportamiento crediticio, como así también su efecto sobre la actividad real. Se explota como cuasi-experimento el anuncio del Banco Central Europeo (BCE), en el primer trimestre del 2018, del cambio de las directrices supervisoras de los préstamos dudosos y sus provisiones. El análisis muestra que, tras la intervención del BCE, los bancos incrementaron el ritmo de cancelación de los préstamos dudosos que habían permanecido un mayor tiempo clasificados como tales. Asimismo, los bancos más expuestos a la intervención endurecieron sus criterios de concesión de préstamos, particularmente hacia las empresas de mayor riesgo, mientras que los bancos con fundamentos más sólidos tuvieron menos reparos en deshacerse de los préstamos dudosos y no fueron tan restrictivos en la concesión de créditos. Por otra parte, el volumen de activos, la inversión, el empleo y las ventas de las empresas con préstamos de los bancos más expuestos a la política del BCE crecieron a un ritmo menor. En su conjunto, los resultados subrayan la importancia de las directrices supervisoras para la gestión de los préstamos dudosos, así como su beneficio potencial para la mejora en la asignación del crédito.

Palabras clave: préstamos dudosos, provisiones para préstamos dudosos, reconocimiento de préstamos dudosos, oferta de crédito, efectos reales sobre empresas.

Códigos JEL: E51, E58, G13, G21.

## 1 Introduction

A common feature of previous financial crises is the accumulation of non-performing loans (NPLs) on banks' balance sheets (Aiyar et al., 2015; Fell et al., 2016; Alessi et al., 2021; Ari et al., 2021; Kasinger et al., 2023). High levels of NPLs bear important consequences for the soundness of the banking sector. They can lead to impaired bank balance sheets, hamper economic growth, and reduce lending capacity (Barseghyan, 2010; Draghi, 2017; Kalemli-Özcan et al., 2022; Charoenwong et al., 2024). Yet, banks may have discretion in their management of NPLs and keep them at inefficiently high levels, because writing them off requires increasing loan loss provisions, which depresses profits in the short run. For this reason, supervisors need to monitor banks' asset quality closely and assess whether they are managing soundly NPLs.

In light of these concerns, the European Central Bank (ECB) undertook specific supervisory initiatives to handle banks' NPLs. Specifically, in March 2018 the ECB published concrete requirements urging banks to increase provisions for NPLs and "comply or explain" the disposal of bad loans. The policy clearly stated supervisory expectations with respect to the age and degree of collateralization of the NPLs. Provisioning guidelines were stricter for uncollateralized NPLs and loans that remained classified as NPLs for longer ("vintage NPL", hereafter). The criteria in the provisioning guidelines were largely unanticipated by the banks. This is evident by negative abnormal returns for Spanish bank stocks around the announcement of the ECB policy.<sup>1</sup>

We use the release of the ECB prudential provisioning for NPLs in March 2018 as a quasi-natural experiment to study whether changes in NPL oversight affect (i) NPL disposals, (ii) bank lending, and (iii) firm outcomes. Our empirical analysis proceeds in three steps. First, to examine the propensity of banks to dispose of NPLs, we utilize loanlevel data from the Spanish Credit Register (CIR), enhanced with bank and firm-level data from regulatory filings and the Central Balance-Sheet Data Office, respectively. We exploit an important feature of the policy that differentially affects NPLs. In particular, the policy prompted banks to increase the provisioning coverage of vintage NPLs. The differences in NPLs vintages allow us to construct a variable gauging whether a loan was affected (i.e., to define a treated and control group, at the loan level), achieving a clear identification of the causal effect of the policy on NPL disposals.

<sup>&</sup>lt;sup>1</sup>See online appendix C for an event study analysis around NPL guidance announcements.

Next, we study how the policy affects banks' lending behavior and the key mechanisms through which banks react to the policy change. For our lending analysis, we employ comprehensive credit information at the bank-firm level from the CIR. To examine whether an increase in vintage NPL disposal matters for banks' credit supply, we compute the weighted average vintage of the NPL in the pre-policy period. This variable measures the degree of banks' exposure to the ECB provisoning expectations. Furthermore, to account for potential time-invariant and time-varying observed and unobserved factors that would potentially bias our results, we employ a comprehensive set of bank-firm and bank controls and gradually introduce a battery of fixed effects.

In the final empirical step, we explore whether the effect of the policy change is transmitted to the real sector. In doing so, we conduct a firm-level analysis to examine whether firms that rely more on lending from banks with higher NPL vintage underperform relative to firms that rely less on such banks. In our analysis, we focus on firms' uptake of bank debt, employment, investment, assets, and sales. Furthermore, we argue that the policy is unlikely to affect all firms in the same manner. In particular, we test whether the effects are stronger for riskier firms, which reveals how the policy change impacts the allocative efficiency in the credit market.

Spain provides a unique setting in which to conduct the empirical analysis for two main reasons. First, in the fourth quarter of 2016, the Spanish banking sector exhibited one of the highest NPL ratios in Europe - around 9% after the sovereign debt crisis of the euro area. Second, the Spanish credit register provides rich loan-level data, which is key for identifying the banks affected by the policy, as well as controlling for a wide range of time-varying and time-invariant firm and bank characteristics that allow identifying how the ECB's policy impacts NPL disposal and bank lending.

We reach a number of novel results that shed light on how the policy intervention affects lending and real outcomes. Our first set of findings suggests that the introduction of the ECB policy affects banks' propensity to dispose of bad loans with older vintages. Compared to the pre-policy period, a 1% increase in the loan vintage doubles the probability of disposing the NPL in the post-policy period (from 1.5 to 3.2 percentage points). Moreover, we find that the propensity to dispose of NPLs after the policy implementation is higher for more profitable banks. These banks are more likely to remove NPLs because they are better positioned to cope with the negative impact of the increase in provisions on profits.

These findings set the stage for our analysis of bank lending. We find that banks more heavily exposed to the policy tighten their lending, thereby suggesting that the policy weakened credit supply. In particular, affected banks decrease lending and require higher levels of collateral in the aftermath of the policy. We also show that banks are more likely to terminate a lending relationship following the policy change. As part of the mechanism, we find that more profitable banks can better sustain lending. Moreover, banks tightened their lending standards more to risky firms, suggesting that the policy discouraged the financing of non-viable (zombie) firms.

Finally, we investigate whether exposure to banks with high NPLs jeopardises financial and real outcomes at the firm level. We document that firms borrowing from banks with higher NPL vintages before the policy experience a decrease in total borrowing, sales, number of workers, investment, and size. More precisely, increasing the value of NPLs by one standard deviation reduces employment and investment growth by 0.7 % and 1.3 %, respectively. This effect is more pronounced for risky firms, implying that the policy had a positive effect on credit allocation in the economy.

We contribute to the literature in four main ways. First, we contribute to the literature on NPL determinants. There is a wide body of research on the determinants of NPLs; it identifies the macroeconomic conditions and bank-specific characteristics as the two main driving factors (Berger and DeYoung, 1997; Balgova et al., 2016). We move this literature forward and analyze the effectiveness of the ECB policy in disposing NPLs. Moreover, we explore whether the link between NPLs and the policy introduction is heterogeneous among different types of NPLs. Consistent with the objectives targeted by the ECB, not all NPLs were affected in a proportional manner, and vintage NPLs were affected the most.

Second, we offer new evidence on how bank supervision affects bank behavior.<sup>2</sup> Asset quality reviews, on-site inspection programs, or guidance and instructions may affect bank decisions. Previous studies find that more intense supervision of exposed banks results in lower risk but is also associated with a reduction in credit supply or slower loan growth, at least in the period immediately following the introduction of the new regime (see Abbassi et al., 2023; Ivanov and Wang, 2023). Moreover, focusing on the role of specific policy

<sup>&</sup>lt;sup>2</sup>For a detailed review, see Hirtle and Kovner (2022).

measures in decreasing NPLs, Accornero et al. (2017) find that banks' lending behavior is not causally affected by the level of NPL ratios. Compared with those papers, our focus is on the ECB intervention and its impact on different types of firms and banks. In doing so, we examine the effect of concrete and stringent provision requirements, as set out in the ECB provisioning requirements.

Third, our paper relates to the literature on the implications of inefficient financing, such as sustained lending to non-viable (zombie) firms. Prior work shows that zombie lending affects the allocation of credit (Blattner et al., 2023; Bonfim et al., 2023). The effective recognition and resolution of NPLs is key to avoid the risk of zombie lending, which has implications for productivity and economic growth (Caballero et al., 2008; Schivardi et al., 2020; Alvarez et al., 2023). In our setting, lending to zombie firms is likely to generate vintage NPLs, thereby increasing the cost of lending to this particular segment of firms. We document that following the policy, banks tend to de-risk by reallocating their resources, as they reduce their lending to firms with high ratios of interest payments to cash inflows.

Finally, our paper is related to the literature analyzing the transmission of disruptions in credit markets to the real economy (see Bonaccorsi and Sette, 2016; Cingano et al., 2016; Bentolila et al., 2018; Farinha et al., 2019; Serena et al., 2022). We show a significant decline in employment, sales, and fixed investment in tangible assets for firms borrowing from banks exposed to the ECB provisioning guidelines. Nonetheless, such effects were particularly acute for risky firms. Therefore, we find that the effects of the supervisory intervention impact on firm-level outcomes, but could have enhanced credit allocation.

The rest of the paper is structured as follows. Section 2 presents an overview of the ECB NPL resolution measures and offers a review of the existing literature. Section 3 contains our data-set description. Section 4 presents our methodology and results. Section 5 provides conclusions and policy implications.

## 2 Institutional background and related literature

## 2.1 ECB's NPL Provisioning Expectations

Supervisors can impact on how banks manage NPLs by taking a number of measures, such as asset quality reviews, on-site inspection programs, or guidance and instructions regarding NPL management. To deal with the significant amount of NPLs in European banks following the euro area sovereign debt crisis, the ECB took a number of measures starting in March 2017. The ECB first released a set of best practices for NPL management, titled "Guidance to Banks on Non-Performing Loans" on March 20, 2017. These measures targeted entities with NPL ratios that are considerably higher than the EU average, which stood at approximately 5% at the end of 2016. The overall objective of this tighter supervisory oversight on NPLs was to enhance the management of NPLs, particularly for these entities.<sup>3</sup>

As the ECB pointed out, the guidance contains predominantly "qualitative elements", and did not make specific recommendations on NPL prudential provisioning. Therefore, to enhance the timeliness of provisions and write-offs, on March 23, 2018 the ECB specified prudential provisioning levels for NPLs, hereafter "ECB NPL Provisioning Expectations."<sup>4</sup> Provisioning expectations depended on the time spent in a non-performing status (i.e., vintage) and its collateral, as shown in Table 1. In particular, NPLs were fully provisioned within two to seven years after being classified as NPLs. The provisioning speed depends on whether loans are secured by collateral, as well as the collateralization ratio. These rules were initially set for new NPLs (classified as such since April 1, 2018). Yet they also provided a supervisory benchmark for the NPL stock, which was explicit in a press release on July 11, 2018.<sup>5</sup>

The ECB NPL Provisioning Expectations applied to significant institutions (SI) within the Single Supervisory Mechanism (SSM, often referred to as ECB Banking Supervision), including their international subsidiaries. These guidelines aimed at tightening the provisioning practices of banks with a larger stock of vintage NPLs (i.e., the loans that have been categorized as NPLs for a long time). They were expected to facilitate NPL disposals for such banks, an issue which in turn would affect their short-term profitability, and ability to originate new loans.<sup>6</sup> Moreover, the ECB NPL Provisioning Expectations

 $<sup>^{3}</sup>$ The ECB policy addresses all non-performing exposures (NPEs) following the EBA definition. Hence, this guidance uses the terms "NPL" and "NPE" interchangeably.

<sup>&</sup>lt;sup>4</sup>See https://www.bankingsupervision.europa.eu/press/pr/date/2019/html/ssm.pr190822~f3dd1be8a4. en.html

<sup>&</sup>lt;sup>5</sup>In such press release, the ECB indicated that it aimed to achieve the same coverage of NPL stock and flow over the medium term. However, it would also consider banks' initial NPL ratios to guide expectations regarding the provisioning coverage of NPL stock.

<sup>&</sup>lt;sup>6</sup>In March 2017 the ECB published a guideline on NPL management, which, however, did not provide quantitative instructions about when and how much to provision NPLs. This could cast doubt on the unexpected nature of the ECB NPL Provisioning Expectations. However, the complex nature of the process and the fact that the draft was not finalized means that banks were unlikely to be able to predict the final policy document accurately, especially with regards to provisioning requirements. We provide evidence in favor of this argument by carrying out an event study analysis in the online appendix. In table A12 we show significant negative stock returns around the time of the ECB's policy communication, lending support to the idea that the NPL provisioning guidelines were largely unanticipated by the market participants.

were relevant for Spanish banks, given the average NPL ratio exhibited at the onset of the policy.

Figure 1 shows the vintage distribution of NPLs for Spain over time. The median vintage decreases, mainly supported by the disposal of NPLs with older vintage. However, the decline in both the median and maximum value of NPL vintage accelerates after the second quarter of 2018 with the introduction of quantitative aspects of supervisory expectations presented in Table 1. As is evident from the discussion thus far, it is fair to assume that Spanish banks were compelled to clean their balance sheets of bad loans following the ECB policy change. In addition, this effect is likely to influence lending and the real economy. It is, therefore, necessary to test these hypotheses using data from the loan market.

#### 2.2 Bank supervision and NPLs: Background literature

Previous literature on NPLs mainly focuses on factors that determine the build-up of bad loans, highlighting the relevance of both macro and micro determinants. For example, adverse macroeconomic conditions associated with sluggish economic activity can boost NPLs through their negative effect on borrowers' wealth, income flows, and debt service capacity (Bernanke et al., 1999; Berger and DeYoung, 1997; Kiyotaki and Moore, 1997). On the banks' side, factors related to profitability, capitalization, or concentration can affect the level of NPLs (Balgova et al., 2016; Bischof et al., 2022).

The high NPL stock is one of the key factors in hindering banks' lending capacity in Europe, and empirical research on the impact of supervisory actions on NPLs is growing. Only a handful of papers use granular data needed to achieve strong identification and establish causality with a higher degree of confidence. Using the AQR in 2014 and data from European banks Bruno and Marino (2018) show that reviewed banks with higher unexpected changes to their NPLs deleverage and reduce their lending more than non-reviewed banks. Likewise, Accornero et al. (2017) use the supervisory intervention associated with the 2014 AQR to show that banks' lending behavior is not causally affected. Using German loan and security data, Abbassi et al. (2023) show that banks reviewed by the ECB in the 2014 AQR reduce their exposure to riskier securities and credit.

Turning to on-site bank inspections as a type of bank supervision, Passalacqua et al. (2021) show that financial intermediaries are more likely to reclassify loans as nonperforming after an audit. Moreover, they reduce lending following the inspection, but this drop reverts to pre-inspection levels after seven quarters. In a similar setting, Bonfim et al. (2023) show that inspections of the largest Portuguese banks reduce zombie lending.<sup>7</sup> Finally, Ivanov and Wang (2023) explore the impact of less lenient supervisors on lending supply, causally linking stricter supervisory evaluations to reduced lending using quasi-random assignment of supervisors to Shard National Credit (SNC) reviews.

These studies provide a helpful background for linking supervisory expectations and bank lending. In this paper, we ask how important a policy intervention regarding NPL provisioning is for NPL disposals, bank lending, and firm-level real outcomes using rich datasets. In the following sections, we turn to our data and estimation strategy.

## **3** Data and descriptive statistics

Our datasets combine information from three sources available at the Bank of Spain: (i) the Spanish Central Credit Register (CIR), (ii) supervisory bank balance sheets and income statements, and (iii) firms' balance sheets from the Spanish Mercantile Register.

The Spanish Central Credit Register contains confidential information on all outstanding loans to non-financial firms granted by all credit institutions operating in Spain. In particular, banks are required to report all loans on their balance sheets to the CIR. It includes loan-level information about the type of loan, amount (drawn and undrawn), type of collateral, maturity, currency, days past due, whether it was forborne or refinanced, the lender, and the borrower. The database also offers borrower-related information, such as firm size. Furthermore, the possibility of identifying both firms and banks enables us to merge the credit register with supervisory bank quarterly balance sheets and annual firm balance sheets, thereby acquiring bank and firm characteristics.

A particular characteristic of the CIR is that it enables us to identify all NPLs (more than 90 days past due) on banks' balance sheets every month. If a bank ceases to report a loan more than 90 days past due in a particular month, we can also identify the disposal of an NPL from the CIR, which might have been due to a write-off or a sale of a loan. This information is of great importance for our analysis on whether the ECB's supervisory provisioning rules for vintage NPLs facilitate the disposal of NPLs from the CIR. We

<sup>&</sup>lt;sup>7</sup>Other approaches by the financial sector to address NPLs include the introduction of asset management companies (Hallerberg and Gandrud, 2015), macroprudential regulation (Cerutti et al., 2015), changes to loan classification, and changes to provisioning stringency (Barth et al., 2004).

consider that a NPL has been disposed if the loan is no longer recorded in the CIR, this way including both NPL sales and write-offs.

Using the CIR, we also compile comprehensive data on the credit exposure of all firms with their respective banks, allowing us to analyze how the ECB's provisioning rules affect bank credit supply. In particular, this bank-firm database allows us to investigate whether banks with higher exposure to vintage NPLs reduce their lending more than banks with lower levels of exposure do, as a result of policies aimed at reducing NPLs from banks' balance sheets.

To analyze firms' real decisions, we combine our credit database with the annual balance sheets of the firms from the Spanish Mercantile Register, which the Central Balance Sheet Office collects. This database permits us to assess whether the ECB's policies affect firms whose loans came primarily from banks with higher NPL vintage. In addition, we follow standard sample selection criteria in the literature and exclude companies with missing data, negative sales, or negative assets.<sup>8</sup>

Table 2 presents the summary statistics for the data. Our analysis centers on Spanish non-financial firms and banks operating in Spain, covering the period 2017q1 to 2019q4. We report figures across the levels of analyses we conduct: loan level (Panel A), bank-firm level (Panel B), bank-level (Panel C), and firm level (Panel D). Over the entire sample, the average NPL disposal rate is 10.8% and the average NPL vintage is approximately three years.

The bank-level summary statistics show that the average value for the ratio of NPLs to total loans is 6.8% and the median is 5.4%. Moving to the liquidity indicator, we observe that the mean liquidity ratio is 10.5%, with a median of 6.6%. Finally, the mean and median values of RoA are 0.47 and 0.51 percent, respectively, possibly reflecting the low profitability, in relative historical terms, in the European banking sector in the aftermath of the euro area debt crisis (Elekdag et al., 2020).

Turning to the firm-level dataset, the average growth rate of bank credit between 2017 and 2019 is 4%. Over the same period, employment increased by 6.6%, investment increased by 7.5%, and total assets increased by 8.5%. Finally, the average firm is well collateralized and liquid, with ratios of 35% and 59%, respectively.

 $<sup>^{8}</sup>$ We winsorize the regression variables at the 1st and 99th percentiles to control for the potential influence of outliers.

## 4 Empirical strategy and results

Our analysis focuses on how the ECB supervisory expectations on bank provisions affect three interrelated outcomes, namely NPL dynamics, bank lending, and firms' financial and real outcomes. In particular, we aim at answering the following questions. First, the extent to which the ECB NPL Provisioning Expectations, released in the first quarter of 2018, affected NPL disposals in Spain. Second, whether bank fundamentals matter for their capacity to dispose of NPLs. Third, whether bank balance sheet cleansing affected banks' ability to provide credit. Finally, whether banks treated the firms the same after the ECB NPL Provisoning Expectations.

It is empirically challenging to identify the direct impact of supervision on both NPLs and loan outcomes, as many observed and unobserved factors can simultaneously drive NPLs, loan outcomes, and supervisory initiatives. For instance, Altavilla et al. (2020) argue that regulatory and disciplinary effects of higher capital ratios can also reduce banks' NPL ratios. Moreover, several factors may jointly determine lending and NPLs at the bank level, such as governance, business model, bank capital, or macroeconomic conditions (Louzis et al., 2012; Hajja, 2020). Therefore, any improvement in bank regulation and/or recovery in economic activity could lead to a negative correlation between NPLs and bank lending. Finally, unobserved shocks at the firm and sector level can affect NPLs and loan dynamics, which requires designing a framework to disentangle these factors from the effect of the policy intervention.

Our research design tackles these concerns in a number of ways. For one, we identify how the policy affects NPL disposals by exploiting the clearly specified quantitative aspects of the ECB's supervisory expectations explained in section 2. This policy incentivizes banks to dispose loans which had been classified as non-performing for a longer period of time (i.e vintage NPLs). In addition, we include a comprehensive set of time-invariant and time-varying fixed effects to isolate our estimates from potential unobserved omitted factors.

When undertaking these analyses, we explore whether the policy's effect on NPL disposals and lending differs among banks with different characteristics. These results shed light on the mechanisms through which the policy affects the supply of lending. We further test whether the policy change affects lending to firms with different degrees of riskiness, which provides evidence about the policy's effect on allocative efficiency in the market for loanable funds.

#### 4.1 ECB NPL Provisioning Expectations and NPL disposals

The ECB NPL Provisioning Expectations set specific and tighter provisioning guidelines on NPLs. This includes a set of quantitative criteria on the timeline for provisioning NPLs dependent on their vintage and collateralization. Specifically, unsecured NPLs required being fully provisioned after two years of classification as an NPL ("vintage"). In contrast, secured NPLs require a certain level of provisioning after three years, increasing until year seven at the latest. As NPL vintage is one of the key determinants of how the ECB policy facilitates the disposal of vintage NPLs, we measure the effectiveness of the policy by testing whether disposals of high vintage NPLs are more likely after the introduction of the policy.

To that aim, we analyze how vintage NPLs affect the propensity of NPL disposal using loan-level information from the second quarter of 2017 to third quarter of 2019, (i.e. five quarters before and after the release of the policy in the second quarter of 2018). The dataset contains 1,654,107 observations from 356,775 loans to 73,422 firms. We track each NPL's presence in the CIR in the subsequent quarter, along with past-due days, outstanding amount, collateral, loan type, and the bank and firm associated with the loan. Specifically, we estimate the following linear probability model:

$$Disposal_{l,b,f,t+1} = \alpha_1 Vint_{l,f,b,t} + \alpha_2 Policy_t \times Vint_{l,f,b,t} + \gamma_{f,t} + \gamma_{b,t} + \gamma_{f,b} + \gamma_{k(l)} + \varepsilon_{l,b,f,t+1}$$
(1)

where  $Disposal_{l,b,f,t+1}$  is a dummy variable that equals 1 if a bank disposes a nonperforming loan l from bank b to firm f, in the next quarter (i.e. t+1), and 0 otherwise.<sup>9</sup>  $Policy_t$  is a dummy variable equal to 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. The loan vintage variable,  $Vint_{l,f,b,t}$ , is the natural logarithm of 1 plus the number of months since the loan was categorized as NPL. Therefore,  $\alpha_1$ measures the probability that an NPL is disposed of the bank's balance sheet in the next quarter in response to a 1% increase in the number of months since the loan became an NPL in the pre-policy period. The coefficient of interest,  $\alpha_2$ , measures the change in how

<sup>&</sup>lt;sup>9</sup>NPLs refer to loans more than 90 days past due. Our definition is consistent with regulatory and supervisory criteria, both at the European and global level (IMF, 2019). For research using this definition see Ari et al. (2021) and Jiménez et al. (2023).

NPL vintage affects disposal probability in the aftermath of the ECB's policy. Therefore,  $\alpha_2 > 0$  implies that the effect of the NPL vintage on the disposal probability increases in the post policy period.

Due to the richness of our dataset, the treatment takes place at the loan level rather than the bank level. This further permits us to control for various loan-level heterogeneity in addition to bank-level heterogeneity that would matter for our results. First, loan recovery rates depend on the degree of collateralization, and the type of collateral pledged. Differences in recovery rates may affect banks' incentives to dispose NPLs backed by collateral and unsecured loans. Similarly, secured loans backed by different types of collateral may also have characteristics, such as liquidity, that cause banks to treat them differently when it comes to disposing of them. Finally, as discussed in section 2, the supervisory expectations of the ECB for vintage NPLs, revealed in detail in July 2018, entail different provisioning rules for different NPL vintages. In our saturated models, we account for such heterogeneity using loan-characteristics fixed effects ( $\gamma_{k(l)}$ ), which is the interaction between a category indicator defining loan types (commercial loans, term loans, credit lines, and leasing), and a category indicator that specifies the collateral type (real estate, financial asset, movable collateral, or uncollateralized), generating 16 categories.

The models include additional controls as follows: bank×quarter fixed effects  $(\gamma_{b,t})$  to account for unobserved time-varying bank heterogeneity, firm×quarter fixed effects  $(\gamma_{f,t})$ to account for time-varying firm heterogeneity and firm-bank fixed effects  $(\gamma_{f,b})$  to account for endogenous matching between firms and banks.  $\varepsilon_{l,b,f,t+1}$  is the error term. Finally, we cluster standard errors at the bank-quarter level to allow for correlation across NPLs from the same bank.

Table 3 presents the results on how the disposal probability of NPLs with different vintages in the next quarter changes before and after the release of the quantitative aspects of the ECB's supervisory expectations. Our key variable of interest is  $Policy_t$  and  $Vint_{l,f,b,t}$  ( $Policy_t \times Vint_{l,f,b,t}$ ). Columns 1 to 3 show that the disposal probability of NPLs with different vintages did not change following the policy change.

In column 4 we control for the time-invariant unobserved characteristics of the bankfirm match. Our results indicate that banks are more likely to dispose of older NPLs in t+1 both before and after the policy. Importantly, banks dispose of older NPLs at a significantly higher rate after the policy shift. To put the numbers into perspective, a 1% increase in the months of a loan classified as an NPL increases the probability that a bank disposes a NPL in t+1 by 1.3 percentage points before the policy change and by 3.2 percentage points after the policy change. In column 5, we show our estimation results for our preferred specification; it saturates the model with the richest set of fixed effects. The estimated coefficient for  $\alpha_1$  suggests that before the ECB's policy, a 1% increase in months classified as an NPL increases the probability of disposal by 1.5 percentage points. There is a significant change after the policy, as the estimated coefficient for  $\alpha_2$  is positive and significant. More important, the estimated effect for the post-policy period, measured as  $\alpha_1 + \alpha_2$ , suggests that a 1% increase in months classified as an NPL increases the disposal probability by 3.2 percentage points.

One potential concern with our findings thus far is that banks may reduce their exposure to vintage NPLs before the ECB initiative. This pattern would violate the paralleltrends assumption, rendering the estimates biased for the effect of the policy. To assess whether any trends before the policy may influence our identification strategy, we investigate the dynamic behavior of our dependent variable over our sample window. Specifically, in Figure 2, we plot the series of coefficients and corresponding 95% confidence intervals from estimating regressions analogous to equation 1, in which we replace  $Policy_t$  with a sequence of time dummies spanning our entire estimation period. The graph suggests that in the run-up to the ECB Provisioning Expectations banks disposed NPLs at a constant rate. The absence of a pre-trend provides more evidence in favor of the causal link between the release of the ECB Provisioning Expectations and the increase in the speed of NPL disposals.

#### 4.1.1 Robustness checks

We conduct a series of robustness tests for the results in the previous sub-section. Table 4 shows the results of the tests. First, we exclude NPLs whose outstanding debt decreases by more than 10% of the last outstanding debt (at its final quarter in the CIR) at any moment before their disposal, aiming at keeping NPLs whose amount remains constant or increases. The results, shown in column 1, are consistent with those in column 5 of Table 3, which is our preferred specification. Yet, they indicate a slightly higher degree of NPL disposals following the policy change.

Second, we estimate the model on the subsample of high NPL banks. The guidance aimed to affect the NPL management of high-NPL banks relative to low-NPL banks. To deal with such a concern, we add bank-quarter fixed effects in our preferred specification, allowing us to exploit NPL vintage variation within the same bank and quarter to explain NPL disposals. Additionally, to better isolate the impact of the provisioning supervisory expectation, we examine separately high NPL banks in the sample, as the ECB guidance should affect them homogeneously. In particular, we drop banks with NPLs below 5%, which is the cut-off the EBA uses to define high-NPL banks. The results of this exercise, presented in column 2, remain consistent with our baseline results.

Third, we make sure that rural banks and foreign credit institutions operating in Spain are not driving our results. Rural banks have a very different business model, relative to other entities. Foreign affiliates may get financial support from their parent banks. To address this potential concern, we rerun our models without rural banks and foreign credit institutions. We report the outcome of this exercise in column 3. We confirm the positive impact of the policy on the rate of NPL disposal.

Finally, we address a potential concern that a particularly vulnerable sector in Spain drives our results. Indeed, the real estate and construction sectors were affected the most during the GFC and the sovereign debt crises and could have contributed to the accumulation of NPLs. Therefore, we re-estimate our regressions excluding the NPLs that belong to firms in the real estate and construction sectors. The results are in column 4 and hold using that sub-sample.

#### 4.1.2 The role of bank-level characteristics

So far we have documented that the NPL vintage has a direct effect on the probability of its disposal after the policy. Now we explore whether certain bank characteristics attenuate or exacerbate this effect. This could be the case as disposal of a high amount of NPLs potentially generates a burden on bank capital and profitability (Altavilla et al., 2018; Elekdag et al., 2020). Therefore, banks' ability to act in line with intended policy outcomes can be closely related to predetermined bank characteristics at the time of the introduction of the policy. In addition, from a broader policy perspective, it is important to assess whether other bank-level factors affect banks' ability to dispose of bad loans and extend new loans in the aftermath of the policy. Further, banks' financial health has been affected significantly by a set of prudential policies in the aftermath of the euro area debt crisis. This provides suggestive evidence of the complementarities between policies to improve bank soundness and policies targeting NPL resolution. In light of this discussion, we take into account how certain bank-level characteristics affect the relationship between NPL vintage and NPL disposal. For this, we estimate equation (1) for our preferred specification (column 5 in Table 3) augmented with interaction terms among the  $Policy_t$ ,  $Vint_{l,f,b,t}$ , and bank-level indicators, such as size, ROA, book capital ratio, and NPL ratio. The results are in Table 5. Each column corresponds to one of the bank-level alternative indicators, with the last column presenting the model with all interaction terms.

Our results show that, following the policy, older vintage level is associated with a higher probability of NPL disposal, especially for more profitable banks. In particular, a one-standard-deviation change in RoA increases the probability of an older NPL being disposed at t+1 by 3.7 percentage points in the aftermath of the ECB policy. This figure implies that, compared to a bank with zero profitability, a bank with profitability one standard deviation higher is approximately twice as much likely to dispose of an older NPL in the next quarter (3.96 percentage points vs 7.7 percentage points, ceteris paribus). We also find that smaller banks are more likely to dispose of NPLs compared to larger banks. For example, compared to a bank that is larger by one standard deviation, following the policy change, a smaller bank's probability of disposing of an NPL in the next period is one percentage point higher. On the other hand, we find that capital ratio and NPL ratio do not influence the propensity to dispose of an NPL following the policy change.

## 4.2 ECB NPL Provisioning Expectations and bank lending

We now aim to understand whether banks with more significant vintage NPLs tightened their lending after the ECB policy. To do so, we use bank-firm level data from the first quarter of 2017 to third quarter of 2019. This dataset includes 6,776,491 observations from 954,097 firm-bank relationships and 305,965 firms. We exploit the data at the firm-bankquarter level and the strategy proposed by Khwaja and Mian (2008) to identify how the ECB policy affects bank credit supply, considering the observed and unobserved factors affecting credit demand. Specifically, we estimate the following equation:

$$y_{f,b,t+1} = \theta_1 Policy_t \times NPL vintage_b + Controls_{f,b,t} + \gamma_{f,t} + \gamma_{f,b} + \varepsilon_{f,b,t+1}$$
 (2)

where the dependent variable  $y_{f,b,t+1}$  can be either (i) the natural logarithm of outstanding credit from bank b to firm f, or (ii) a dummy that equals 1 if bank b extends a new credit to firm f and 0 otherwise, or (iii) a dummy that equals 1 if bank b terminates the lending relationship with firm f and 0 otherwise, or (iv) the ratio of collateralized credit that firm f has with bank b, or (v) the ratio of credit with residual maturity beyond three years that firm f has with bank b.

The exposure of the bank to the policy is represented by NPL vintage, which is the weighted average of NPL vintage (number of months since the loan became an NPL) of loans by bank b to non-financial firms as the end of the fourth quarter of 2017. More concretely, we define NPL vintage as:

$$NPL \ vintage_b = \frac{\sum_{l=1}^{N_b} vintage_{l,b} \times C_{l,b}}{\sum_{l=1}^{N_b} C_{l,b}}$$
(3)

In this expression, the NPL vintage of each loan in the fourth quarter of 2017 is weighted by its outstanding credit (i.e.,  $C_{l,b}$ ), in the same period across all the loans in the loan portfolio of bank b (i.e.,  $l \in [1, 2, 3, ..., N_b]$ ). By definition, performing loans have a vintage of zero. Importantly, we additionally control for the bank's NPL ratio in our set of bank controls, allowing us to isolate the effect of the bank's NPL vintage from the NPL levels. Thus, banks with more significant amounts of vintage NPL before the policy have higher NPL vintage values, making them more sensitive to the ECB NPL Provisioning Expectations. We also control for a vector of bank-firm level factors denoted by  $Controls_{f,b,t}$ , which are lagged to limit endogeneity concerns. At the bank-firm level, the vector includes the share of NPL, collateralized loans, forborne/refinanced loans, long-term loans to firm f from bank b, and the ratio of loans from bank b to firm f total debt to banks. We also add bank-level controls such as the logarithm of total assets, ROA, NPL, liquidity, and leverage ratios. Additionally, we include firm-time fixed effects to control for all (un)observed heterogeneity (firm-level credit demand, firm quality, growth opportunities, riskiness, etc.). This is particularly important to give a supply-side interpretation of the effects: the inclusion of firm-time fixed effects ensures we are comparing the same borrower with at least two different banks and, therefore, absorbing any demand factors.

In addition, we saturate our models with bank-firm fixed effects to control for a persistent (non-random) firm-bank specific match, such as geographical distance and relationship lending (Petersen and Rajan, 1995). Finally, we add bank type-quarter fixed effects to account for different shocks that could have distinctly affected different types of banking institutions, which our set of bank controls does not capture. In particular, we consider three types of banking institutions: banks directly supervised by the ECB, rural banks, and other banks.

The coefficient  $\theta_1$  indicates the extent to which banks with different levels of vintage NPLs reduce their lending standards to the same borrower, following the ECB policy. Given that the policy caused a decrease in NPL balances, we only consider firms with performing credit greater than zero to avoid a mechanical decay in our credit variables. We cluster standard errors at the bank-quarter level.

Table 6 shows the estimation results for the model presented in equation 2. Columns 1 to 5 show the results when considering the natural logarithm of credit, the new credit dummy, the termination dummy, the share of collateralized credit, and the share of long-term credit, respectively. The general finding is that banks with higher levels of vintage NPL tighten lending standards more than other banks. According to column 1, a one-standard-deviation increase in NPL vintage, which roughly corresponds to 1.9 months, reduces credit by 2.7% (-0.0061×4.503). These findings echo Abbassi et al. (2023), who show that credit supply is lower for banks subject to the ECB's asset quality review (AQR). Also, Altavilla et al. (2020) find that supranational supervision reduces credit supply to firms with high ex-ante and ex-post credit risk. We further document that the probability of ending a lending relationship increases by 0.32 percentage points, and the tendency to collateralize loans increases by 0.78 percentage points in the aftermath of the ECB policy. On the other hand, we do not find a significant change in the probability of obtaining a new loan or an increase/decrease in the share of long-term loans.<sup>10</sup>

As in the case of NPL disposals, Figure 3 shows no trends before the introduction of the policy for our main specifications. This supports our identifying assumption that the banks did not anticipate the details of the policy regarding the quantitative aspects of the supervisory expectations on NPL disposal. This figure also provides visual evidence that the effect of the policy change on the credit occurred with a time lag, with the effect increasing over time.

In sum, the results are economically important and suggest that affected banks curtail the supply of credit after the ECB policy. Hence, we postulate that more exposed banks

<sup>&</sup>lt;sup>10</sup>We do not find a drop in lending for banks with higher balances of non-performing loans during the period around the release of the ECB Guidelines (2017q1).

were forced to recognize risky loans and increase loan disposal, thereby creating pressure on their lending capacity.

#### 4.2.1 Robustness checks

In Table 7, we present five additional robustness tests for the results in column (1) of Table 6. We first allow for fixed effects for the group of banks that participated in the 2018 EU-wide stress test. In particular, the stress test might cause less sound banks to deleverage to preserve capital for the test. Concerns are accentuated by the fact that only the four largest Spanish banks participated, showing significant resilience against the adverse stress scenario. To address this issue, in column 1 we include a bank group dummy (1 if the bank took part in the stress test, and 0 otherwise) interacted with quarter dummies to deal with the possibility of different trends between participating and non-participating banks in the 2018 EU-wide stress test. In column 2, to avoid concerns regarding the endogeneity of the control variables, we measure them in the pre-policy period and then interact them with time fixed effects to allow a differential impact around the policy shift. As in our analysis of NPL disposals, we also conduct the following robustness checks: we drop from the sample banks with NPL ratios below 5% (column 3); we exclude rural banks, and foreign credit institutions operating in Spain (column 4); and we remove firms belonging to the construction or real estate sectors (column 5). Our findings are robust to all the above modifications.

#### 4.2.2 The role of bank characteristics

At the next step, we explore whether bank characteristics such as profitability, NPL ratios, NPL coverage ratios, size, and capital, affect the degree to which bank lending standards respond to the policy change. For this, we estimate the model in Equation 2 but include the interaction terms among NPL vintage, the policy dummy, and the various bank-level characteristics.

We first find that the banks at the mean of the corresponding distributions of these bank characteristics do not experience any change in lending behavior. However, we bring into the surface significant heterogeneities across banks, which also reveals important insights into how the policy affects the credit market.

Our findings in general reveal that the banks with weak fundamentals do reduce lending following the tighter policy, which requires disposal of vintage NPLs. For example, consistent with our analysis of NPL disposals, Table 8 shows that bank profitability is one of the factors that interact with the effect of the policy on bank lending. In particular, the policy negatively affects lending by banks with low profitability. For example, a bank with profitability two standard deviations lower than the average bank significantly reduces lending compared to the latter, assuming their NPLs exhibits the same vintage. We also find that bank asset affects how bank lending responds to the tighter NPL provisioning requirements. In particular, banks with higher NPL ratios and lower NPL coverage ratios than the average bank reduce their lending following the policy. For example, keeping everything else constant, for a bank with an NPL coverage ratio below two standard deviations lower than the average, the effect of the NPL provisioning policy on bank lending turns contractionary. In other words, banks with a higher share of NPLs that are not provisioned by the time of the policy reduce their lending in response to tighter provisioning requirements. Finally, we interpret the coefficient associated with the size interaction as indicative of heightened supervisory scrutiny on larger banks. For instance, the BIS core principles emphasize that the supervisory intensity applied to a bank should consider its systemic importance.

These results highlight that strengthening banks' financial health in the aftermath of a crisis may be a pre-condition to guarantee a prompt resolution of NPLs without significantly disrupting financial intermediation. Thus, the success/failure of regulatory and supervisory efforts aiming to improve banks' asset quality may depend on the initial soundness of the banking system.

## 4.3 ECB's NPL Provisioning Expectations and firm-level outcomes

Thus far, we have shown that Spanish banks with more significant vintage NPLs cut credit to their borrowers more after the ECB NPL Provisioning Expectations. In this section, we first reaffirm the decline in credit volume from the firms' perspective. We then explore the impact of the policy on firms' real outcomes, such as growth of employment, investment, total assets and, sales.

The firm-level analysis requires us to match our previous dataset with the Spanish Mercantile Register to obtain firm balance-sheet variables. As a consequence, our sample size decreases to 113,081 firms (38.6% of the original sample), particularly biased toward larger firms (representing 51% of the outstanding debt at the end of 2017).<sup>11</sup> For this analysis, we estimate the following empirical model:

$$Firm \ Outcome_{17:19,f} = \theta_1 Weighted \ NPL \ vintage_{17,f} + Controls_{17,f} + \gamma_{P,I,Size} + \varepsilon_f,$$
(4)

where the dependent variable *Firm Outcome* measures a firm's real variable growth between the end of 2017 and 2019 (i.e. before and after the introduction of the ECB policy). As the outcomes of interest, we focus on the growth in banks' credit commitments including both drawn and undrawn amounts, number of employees, tangible fixed assets, total assets, and sales. Because we focus on the firm level, we construct the exposure of firm fto its banks with different levels of vintage NPLs as:

Weighted NPL 
$$vintage_{17,f} = \sum_{b=1}^{N_f} w_{17,f,b} NPL \ vintage_b$$
 (5)

where Weighted NPL vintage is the weighted average NPL vintage for each firm, considering the NPL vintage of each bank b and the share of loans from each bank in the total outstanding bank debt of firm f. In particular, it uses weights,  $w_{17,f,b}$ , equal to the amount each bank b lends to firm f considering the number of all banks in a lending relation with firm f (i.e.,  $N_f$  by 2017). Thus, a higher NPL vintage indicates that firm fhas a higher exposure to banks with high-vintage NPLs, as its lending mainly comes from such institutions.

Our parameter of interest is  $\theta_1$ , which captures the extent to which banks with high NPL vintage affect firm-level outcomes for the average firm. We control for a set of variables for accommodating sources of credit and investment/growth opportunities measured as of the end of 2017. First, we include firm-level characteristics such as size, book-to-capital ratio, liquidity ratio, ROA, tangible assets to total assets, age, and riskiness (measured as the share of credit that is either forborne or NPL). In addition, we control for bank-level characteristics, such as the logarithm of bank assets, bank capital ratio, bank liquidity ratio, ROA, and share of lending from significant institutions weighted by credit from each bank to firm f. We also include bank-firm contractual characteristics,

<sup>&</sup>lt;sup>11</sup>Despite the drop in the sample of firms, results are similar to those employing the entire sample. Table D.1 in the Appendix replicates the results of Table 6 for this sub-sample.

such as the share of long-term debt, the share of collateralized debt, and the share of credit from the main bank. Finally, we control for province-industry-size fixed effects denoted by  $\gamma_{P,I,Size}$ . We cluster standard errors at the main bank level.

The results of estimating equation 5 are in Table 9. In column 1 we find a reduction in the growth rate of bank debt for firms that rely more heavily on exposed banks. In particular, a one-standard-deviation increase in NPL vintage decreases the growth rate of committed loans by 2.3 percentage points. These results also indicate that the average firm is not capable of smoothing the decrease in credit supply, which complements our earlier results from the banks' perspective. In other words, the higher value of vintage NPLs before the policy is associated with a decline in bank lending in the aftermath of the ECB intervention.

According to columns 2 to 5, lower access to credit following the policy has implications for firms' real decisions. For instance, we show that more exposed firms (a one-standarddeviation increase in the weighted NPL vintage) experience 0.7% lower employment growth if they obtain loans from more exposed banks. The results also document a deterioration in firms' investment in tangible assets by 1.3%. Finally, we find that firms' size (proxied by total assets) and sales grew at a slower rate following the policy if their lending comes from banks with higher exposure to the policy. Our results suggest that firms' bank debt and outcomes respond to changes in the vintage of the NPLs for firms borrowing from banks exposed to the ECB policy shift.

#### 4.3.1 Firm riskiness

In order to enrich our findings about the supply of credit and firm-level outcomes, we now turn our attention to the role of firm riskiness. Specifically, we check whether borrowing from an exposed bank has a stronger effect on firms that are riskier in the pre-policy period. One of the potential outcomes associated with the supervisory and regulatory policy interventions is the change in banks' risk exposure in the aftermath of the policy change. In particular, by introducing tighter provisioning requirements for NPLs, policymakers may increase banks' incentives to reduce credit risk exposure in the aftermath of the policy change. This can result in higher attention for banks to lending relationships that carry higher risk of becoming NPLs in the future. These may induce a higher reduction in lending to risky firms and/or asking for higher collateral. To this end, we interact our variables of interest in equation (2) with a variable measuring firm riskiness. In the spirit of Gertler and Gilchrist (1994) and Yang et al. (2022), *Risky* is a dummy variable that equals 1 if the firm's interest coverage ratio is above the median of the distribution of the variable as of the end of 2017. This exercise is based on the hypothesis that the banks may implement more strict credit standards for more risky firms compared to less risky firms following the policy change. The estimation results in column 1 of Table 10 show that banks with higher exposure to policy due to larger stocks vintage NPLs cut lending after the policy change. Notably, this effect is more potent if a firm is risky. To put the numbers into perspective, we find that after the policy, a bank with a one-standard-deviation increase in NPL vintage cut lending to non-risky firms by 1.08%. On the other hand, the effect on the risky firms is statistically significant, with a 1.6% decline.

In terms of the mechanism, we next ask whether this decline stems from a drop in the origination of new loans to risky firms, an increase in the termination of banking relationships with risky firms, or both. Results in column 2 of Table 10 show that banks decrease their tendency to originate new loans to risky firms following the policy, whereas less-risky firms, defined as the firms with interest coverage ratio below the median of the distribution, do not experience a decline in the access to new loans. However, column 3 suggests that risky and non-risky firms do not face a significant change in the probability of their lending relationship being terminated.

Column 4 of Table 10 presents further analysis on whether banks increase collateral requirements when lending to risky and non-risky firms. In particular, the collateralized loan ratio increases only for the risky firms after the policy change. Finally, column 5 shows that the long-term loan ratio decreases both for risky and non-risky firms, where the latter faces a higher decline. Thus, in the aftermath of the policy, risky firms primarily received shorter-maturity loans and were required to pledge more collateral when borrowing from banks with higher exposure to the policy.

As the last exercise, we assess the differential role of firm riskiness on firm-level financial and real outcomes. In particular, using equation 4, we focus on whether borrowing from exposed banks affects financial choices and real outcomes of risky and non-risky firms differently. The results, reported in Table 11, show that risky and non-risky firms differ in borrowing outcomes significantly. However, the effect on the former group is significantly higher than the effect on the latter. Moreover, risky firms perform significantly worse than non-risky firms in terms of employment and investment growth. Depending on the specification, we find that the rate of slowdown in employment, associated with exposure to affected banks in the aftermath of the ECB's policy, is two-to-four times higher for risky firms. Likewise, following the policy, risky firms exposed to affected banks experience a slowdown in investment growth at a rate about two-to-three times higher than their nonrisky counterparts.

All in all, we find that the policy brings about an improvement in allocative efficiency in the economy. In particular, we observe emergence of derisking behavior for banks exposed to the new regulations concerning NPLs, as they have a significantly lower tendency to originate loans to risky firms after the introduction of the ECB's supervisory expectations.

## 5 Conclusion

We use the release of the ECB NPL Provisioning Expectations in 2018 as a quasi-natural experiment to study how changes in NPL oversight affect (i) NPL dynamics and disposals, (ii) bank lending, and (iii) firm dynamics. For our analysis of NPL disposals, we use loan-level data from the Spanish Credit Register (CIR), enhanced with bank and firm-level data from regulatory filings and the Central Balance-Sheet Data Office, respectively. Our findings indicate that the ECB supervisory measures trigger a reduction in bank NPLs. We also show that banks with better bank fundamentals dispose of their NPLs at a higher rate.

This policy had impact on lending. We find that banks with higher levels of vintage NPLs reduce lending more in the aftermath of the policy. In other words, lending from exposed banks was reduced after the policy, thereby suggesting bank-driven tightening in the credit market due to the policy. In addition, banks with higher profitability and asset quality at the date of the release are more capable of smoothing the effect of the policy change on the credit market. Equally important, the credit tightening was particularly focused on risky firms, indicating that the policy discouraged banks from financing zombie firms.

Finally, we show that firms exposed to high-vintage banks experience a decline in borrowing, cut employment, and reduce their investment in fixed assets. Moreover, we observe that the reduction in bank debt and employment is more pronounced for the risky firms. All in all, we conclude that the ECB's policy is effective in reducing NPLs. Yet, this was associated with a decline in bank lending in the short term as well as real effects, which is consistent with recent papers focusing on improving asset quality at banks. In addition, the policy affected the most risky firms, possibly improving credit allocation in the economy, because risky firms are more likely to generate vintage NPLs. We acknowledge that an analysis of the medium- and long-run effects of supervisory actions is beyond the scope of this paper, as the period between the first quarter of 2020 and afterwards are associated with the Covid-19 pandemic.

## References

- Abbassi, Puriya, Rajkamal Iyer, José-Luis Peydró and Paul Soto. (2023). "Stressed banks? Evidence from the largest-ever supervisory review". *Management Science*, forthcoming.
- Accornero, Matteo, Piergiorgio Alessandri, Luisa Carpinelli and Alberto Maria Sorrentino. (2017). "Non-performing loans and the supply of bank credit: Evidence from Italy". Working Paper, 374, Bank of Italy. https://www.bancaditalia.it/pubblicazioni/qef/2017-0374/QEF\_374.pdf
- Aiyar, Shekhar, Wolfgang Bergthaler, José Garrido, Anna Ilyina, Andreas Jobst, Kenneth Kang, Dmitriy Kovtunand, Yan Liu, Dermott Monaghan and Marina Moretti. (2015). "A strategy for resolving Europe's problem loans". Discussion note, International Monetary Fund. https://www.imf.org/external/pubs/ft/sdn/2015/sdn1519.pdf
- Alessi, Lucia, Brunella Bruno, Elena Carletti, Katja Neugebauer and Isabella Wolfskeil. (2021). "Cover your assets: non-performing loans and coverage ratios in Europe". *Economic Policy*, 36(108), pp. 685-733. https://doi.org/10.1093/epolic/eiab013
- Altavilla, Carlo, Miguel Boucinha, José-Luis Peydró and Frank Smets. (2020). "Banking supervision, monetary policy and risk-taking: Big data evidence from 15 credit registers". Working Paper, 2349, European Central Bank. https://www.ecb.europa.eu/pub/pdf/scpwps/ ecb.wp2349~515abecc84.en.pdf
- Altavilla, Carlo, Miguel Boucinha and José-Luis Peydró. (2018). "Monetary policy and bank profitability in a low interest rate environment". *Economic Policy*, 33(96), pp. 531-586. https://doi.org/10.1093/epolic/eiy013
- Álvarez, Laura, Miguel García-Posada and Sergio Mayordomo. (2023). "Distressed firms, zombie firms and zombie lending: A taxonomy". *Journal of Banking and Finance*, 149(106762). https://doi.org/10.1016/j.jbankfin.2023.106762
- Ari, Anil, Sophia Chen and Lev Ratnovski. (2021). "The dynamics of non-performing loans during banking crises: A new database with post-COVID-19 implications". *Journal of Banking and Finance*, 133(106140). https://doi.org/10.1016/j.jbankfin.2021.106140
- Balgova, Maria, Michel Nies and Alexander Plekhanov. (2016). "The economic impact of reducing non-performing loans". Working Paper, 193, European Bank for Reconstruction and Development. https://www.ebrd.com/publications/working-papers/economic-impact. html
- Barseghyan, Levon. (2010). "Non-performing loans, prospective bailouts, and Japan's slowdown". *Journal of Monetary Economics*, 57(7), pp. 873-890. https://doi.org/10.1016/j.jmoneco.2010.08.002
- Barth, James, Gerard Caprio and Ross Levine. (2004). "Bank regulation and supervision: What works best?". *Journal of Financial Intermediation*, 13, pp. 205-248. https://doi.org/10.1016/j. jfi.2003.06.002
- Bentolila, Samuel, Marcel Jansen, Gabriel Jiménez and Sonia Ruano. (2018). "When credit dries up: Job losses in the great recession". *Journal of the European Economic Association*, 16, pp. 650-695. https://doi.org/10.1093/jeea/jvx021

- Berger, Allen, and Robert DeYoung. (1997). "Problem loans and cost efficiency in commercial banks". *Journal of Banking and Finance*, 21, pp. 849-870. https://doi.org/10.1016/S0378-4266(97)00003-4
- Bernanke, Ben, Mark Gertler and Simon Gilchrist. (1999). "The financial accelerator in a quantitative business cycle framework". In J. Taylor, and M. Woodford (eds.), *Handbook of Macroeconomics*, vol. 1C. Elsevier Science, North Holland, pp. 1341-1393. https://doi.org/10.1016/S1574-0048(99)10034-X
- Bischof, Jannis, Nicholas Rudolf and Wilhelm Schmundt. (2022). "How do non-performing loans evolve along the economic cycle? The role of macroeconomic conditions and legal efficiency". *European Accounting Review*, 31(5), pp. 1149-1174. https://doi.org/10.1080/09638 180.2022.2071960
- Blattner, Laura, Luisa Farinha and Francisca Rebelo. (2023). "When losses turn into loans: The cost of weak banks". *American Economic Review*, 113(6), pp. 1600-1641. https://doi. org/10.1257/aer.20190149
- Bonaccorsi, Emilia, and Enrico Sette. (2016). "Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register". *Journal of Financial Intermediation*, 25, pp. 54-76. https://doi.org/10.1016/j.jfi.2015.06.002
- Bonfim, Diana, Geraldo Cerqueiro, Hans Degryse and Steven Ongena. (2023). "On-site inspecting zombie lending". *Management Science*, 69, pp. 2547-3155. https://doi.org/10.1287/mnsc.2022.4452
- Bruno, Brunella, and Immacolata Marino. (2018). "How do banks respond to non- performing loans?". Working Paper, 13, Centre for Studies in Economics and Finance, University of Naples. https://www.csef.it/WP/wp513.pdf
- Caballero, Ricardo, Takeo Hoshi and Anil Kashyap. (2008). "Zombie lending and depressed restructuring in Japan". *American Economic Review*, 98, pp. 1943-1977. https://doi.org/10.1257/aer.98.5.1943
- Cerutti, Euginio, Stijn Claessens and Luc Laeven. (2015). "The use and effectiveness of macroprudential policies: New evidence". Working Paper, 15/61, International Monetary Fund. https://www.imf.org/external/pubs/ft/wp/2015/wp1561.pdf
- Charoenwong, Ben, Meng Miao and Tianyue Ruan. (2024). "Nonperforming loan disposals without resolution". *Management Science*, forthcoming. https://doi.org/10.1287/mnsc.2022.02163
- Cingano, Federico, Francesco Manaresi and Enrico Sette. (2016). "Does credit crunch investments down? New evidence on the real effects of the bank-lending channel". *Review of Financial Studies*, 29(10), pp. 2737-2773. https://doi.org/10.1093/rfs/hhw040
- Correa, Ricardo, Kuan-Hui Lee, Horacio Sarpiza and Gustavo A. Suárez. (2014). "Sovereign credit risk, banks' government support, and bank stock returns around the world". Journal of Money, *Credit and Banking*, 46, pp. 93-121. https://doi.org/10.1111/jmcb.12080
- Draghi, Mario. (2017). "European banking supervision three years on". Welcome remarks by Mario Draghi, president of the ECB, at the second ECB forum on banking supervision, Frankfurt am Main. https://www.ecb.europa.eu/press/key/date/2017/html/ecb.sp171107.en.html

- Elekdag, Selim, Sheheryar Malik and Srobona Mitra. (2020). "Breaking the bank? A probabilistic assessment of euro area bank profitability". *Journal of Banking and Finance*, 120(105949). https://doi.org/10.1016/j.jbankfin.2020.105949
- Farinha, Luisa, Marina-Eliza Spaliara and Serafeim Tsoukas. (2019). "Bank funding shocks and firm performance: New evidence from the sovereign debt crisis". *Journal of Financial Intermediation*, 40(100818). https://doi.org/10.1016/j.jfi.2019.01.005
- Fell, John, Maciej Grodzicki, Dejan Krušec, Reiner Martin and Edward O'Brien. (2016). "Overcoming non-performing loan market failures with transaction platforms". Special features, European Central Bank. https://www.ecb.europa.eu/pub/pdf/other/ecb. sfafinancialstabilityreview201711.en.pdf
- Gertler, Mark, and Simon Gilchrist. (1994). "Monetary policy, business cycles, and the behavior of small manufacturing firms". *Quarterly Journal of Economics*, 109, pp. 309-340. https://doi.org/10.2307/2118465
- Hajja, Yaman. (2020). "Impact of bank capital on non-performing loans: New evidence of concave capital from dynamic panel-data and time series analysis in Malaysia". *International Journal of Finance and Economics*, 27(3), pp. 2921-2948. https://doi.org/10.1002/ijfe.2305
- Hallerberg, Mark, and Christopher Gandrud. (2015). "Building better bad banks: The political economy of public asset management companies for resolving financial crises". Tech. rep., Mimeo. https://fgch.github.io/amc-site/
- Hirtle, Beverly, and Anna Kovner. (2022). "Bank supervision". *Annual Review of Financial Economics*, 14, pp. 39-56. https://doi.org/10.1146/annurev-financial-111620-022516
- International Monetary Fund. (2019). "Financial soundness indicators". Compilation guide, International Monetary Fund. https://www.imf.org/en/Data/Statistics/FSI-guide
- Ivanov, Ivan, and James Wang. (2023). "Bank supervision and corporate credit supply". *Management Science*, 70. https://doi.org/10.1287/mnsc.2023.4854
- Jiménez, Gabriel, Luc Laeven, David Martínez-Miera and José-Luis Peydró. (2023). "Public guarantees, relationship lending and bank credit: Evidence from the covid-19 crisis". Working Paper, 2318, Banco de España. https://doi.org/10.53479/30812
- Kalemli-Özcan, Şebnem, Luc Laeven Luc and David Moreno. (2022). "Debt overhang, rollover risk, and corporate investment: Evidence from the european crisis". *Journal of the European Economic Association*, 20, pp. 2353-2395. https://doi.org/10.1093/jeea/jvac018
- Kasinger, Johannes, Jan Pieter Krahnen, Steven Ongena, Loriana Pelizzon, Maik Schmeling and Mark Wahrenburg. (2023). "Non-performing loans - new risks and policies? NPL resolution after COVID-19: Main differences to previous crises". Working Paper, 84, SAFE. https://www. europarl.europa.eu/RegData/etudes/STUD/2021/651387/IPOL\_STU(2021)651387\_EN.pdf
- Khwaja, Asim, and Atif Mian. (2008). "Tracing the impact of bank liquidity shocks: Evidence from an emerging market". *American Economic Review*, 98, pp. 1413-1442. https://doi.org/ DOI:10.1257/aer.98.4.1413
- Kiyotaki, Nobuhiro, and John H. Moore. (1997). "Credit cycles". *Journal of Political Economy*, 105, pp. 211-248. https://doi.org/10.1086/262072

- Louzis, Dimitrios, Angelos Vouldis and Vasilios L. Metaxas. (2012). "Macroeconomic and bankspecific determinants of non-performing loans in greece: A comparative study of mortgage, business and consumer loan portfolios". *Journal of Banking and Finance*, 36(4), pp. 1012-1027. https://doi.org/10.1016/j.jbankfin.2011.10.012
- Passalacqua, Andrea, Paolo Angelini, Francesca Lotti and Giovanni Soggia. (2021). "The real effects of bank supervision: Evidence from on-site bank inspections". Working Paper, 1349, Bank of Italy. https://www.bancaditalia.it/pubblicazioni/temi-discussione/2021/2021-1349/en\_tema\_1349.pdf
- Petersen, Mitchell, and Raghuram Rajan. (1995). "The effect of credit market competition on lending relationships". *Quarterly Journal of Economics*, 110, pp. 407-443. https://doi. org/10.2307/2118445
- Schivardi, Fabio, Enrico Sette and Guido Tabellini. (2020). "Identifying the real effects of zombie lending". *Review of Corporate Finance Studies*, 9, pp. 569-592. https://doi.org/10.1093/rcfs/ cfaa010
- Serena, José-María, Marina-Eliza Spaliara and Serafeim Tsoukas. (2022). "International bank lending and corporate debt structure". *Economic Inquiry*, 60, pp. 1214-1232. https://doi. org/10.1111/ecin.13078
- Yang, Junhong, Alessandra Guariglia, Yuchao Peng and Yukun Shi. (2022). "Inventory investment and the choice of financing: Does financial development play a role?" *Journal of Corporate Finance*, 74(102139). https://doi.org/10.1016/j.jcorpfin.2021.102139

## A Figures and tables

Vintage	Unsecured part	Secured part
2 years	100%	
3 years		40%
4 years		55%
5 years		70%
6 years		85%
7 years		100%

 Table 1: ECB's quantitative expectations on NPL provisioning

Source: European Central Bank (ECB)



Figure 1: NPL vintage distribution over time

	Obs.	Mean	$\operatorname{sd}$	p25	Median	p75
Disposal	$1,\!654,\!107$	0.108	0.311	0.000	0.000	0.000
Policy	$1,\!654,\!107$	0.501	0.500	0.000	1.000	1.000
Loan size	$1,\!654,\!107$	8.463	2.905	6.265	8.798	10.784
Log(1+Vintage)	$1,\!654,\!107$	2.969	1.064	2.303	3.105	3.804
Real estate collateral	$1,\!654,\!107$	0.253	0.435	0.000	0.000	1.000
Financial asset collateral	$1,\!654,\!107$	0.006	0.078	0.000	0.000	0.000
Movable collateral	$1,\!654,\!107$	0.004	0.062	0.000	0.000	0.000
Uncollateralized	$1,\!654,\!107$	0.737	0.440	0.000	1.000	1.000
Commercial loan	$1,\!654,\!107$	0.046	0.210	0.000	0.000	0.000
Leasing	$1,\!654,\!107$	0.039	0.193	0.000	0.000	0.000
Credit line	$1,\!654,\!107$	0.469	0.499	0.000	0.000	1.000
Term loans	$1,\!654,\!107$	0.446	0.497	0.000	0.000	1.000
<b>Panel B</b> : Bank-firm-level variables						

## **Panel A**: NPL-level variables

	Obs.	Mean	$\operatorname{sd}$	p25	Median	p75
Log(Credit)	6,776,491	1 10.897	7 2.014	9.703	10.988	12.164
New credit dummy	6,776,491	1 0.217	0.412	0.000	0.000	0.000
Termination dummy	8,284,342	2 0.039	0.195	0.000	0.000	0.000
Collateralized loan share	6,776,491	1 0.189	0.372	0.000	0.000	0.000
Long-term loan share	6,776,491	1 0.644	0.418	0.161	0.928	1.000
NPL share	6,776,491	1 0.007	0.080	0.000	0.000	0.000
Forborne loan share	6,776,491	1 0.021	0.133	0.000	0.000	0.000
Credit share	6,776,491	1 0.331	0.285	0.089	0.249	0.522
<b>Panel C</b> : Bank-level variables						
Panel C: Bank-level variables	Obs.	Mean	sd	p25	Median	p75
Panel C: Bank-level variables Log(Assets)	Obs. 1,051	Mean 14.296	sd 2.150	p25 12.625	Median 14.135	p75 15.309
Panel C: Bank-level variables Log(Assets) Capital ratio	Obs. 1,051 1,051	Mean 14.296 7.017	sd 2.150 14.734	p25 12.625 5.658	Median 14.135 7.660	p75 15.309 9.685
Panel C: Bank-level variables Log(Assets) Capital ratio NPL ratio	Obs. 1,051 1,051 1,051	Mean 14.296 7.017 6.879	sd 2.150 14.734 7.185	p25 12.625 5.658 2.734	Median 14.135 7.660 5.473	p75 15.309 9.685 8.151
Panel C: Bank-level variables Log(Assets) Capital ratio NPL ratio Liquidity ratio	Obs. 1,051 1,051 1,051 1,051	Mean 14.296 7.017 6.879 10.478	sd 2.150 14.734 7.185 13.189	p25 12.625 5.658 2.734 2.793	Median 14.135 7.660 5.473 6.604	p75 15.309 9.685 8.151 11.876
Panel C: Bank-level variables Log(Assets) Capital ratio NPL ratio Liquidity ratio ROA	Obs. 1,051 1,051 1,051 1,051 1,051	Mean 14.296 7.017 6.879 10.478 0.477	sd 2.150 14.734 7.185 13.189 1.080	p25 12.625 5.658 2.734 2.793 0.282	Median 14.135 7.660 5.473 6.604 0.512	p75 15.309 9.685 8.151 11.876 0.771

Table 2 (cont'd)	: Summary	<sup>r</sup> statistics
------------------	-----------	-------------------------

	Obs.	Mean	$\operatorname{sd}$	p25	Median	p75
Bank credit growth	113,081	0.040	0.926	-0.313	-0.023	0.357
Employment growth	$113,\!081$	0.066	0.477	-0.063	0.025	0.211
Investment growth	$113,\!081$	0.075	0.718	-0.152	-0.015	0.214
Firm growth	$113,\!081$	0.085	0.310	-0.066	0.043	0.206
Sales growth	$113,\!081$	0.081	0.345	-0.067	0.074	0.233
Weighted NPL vintage	$113,\!081$	3.757	1.892	2.153	3.692	5.159
Risky	$113,\!081$	0.502	0.500	0.000	1.000	1.000
Log(Assets)	$113,\!081$	6.686	1.607	5.573	6.501	7.611
Capital ratio	$113,\!081$	0.353	0.278	0.160	0.339	0.550
Liquidity ratio	$113,\!081$	0.588	0.281	0.369	0.626	0.834
ROA	$113,\!081$	0.031	0.078	0.004	0.020	0.056
Tangible assets to total assets	$113,\!081$	0.349	0.272	0.111	0.293	0.545
$\log(1+age)$	$113,\!081$	2.663	0.760	2.303	2.833	3.178

Panel	$\mathbf{D}$ :	Firm-level	variables
-------	----------------	------------	-----------

The table provides basic descriptive statistics. *Disposal* is a dummy variable that takes value one if the bank stopped reporting the non-performing loan (NPL) to the CIR. *Policy* is a dummy variable that takes value one after the release of the ECB Provisioning Guidelines for NPLs. *Vintage* is the number of months that the loan is classified as NPL. NPL *vintage* is the weighted average vintage of a bank's loan portfolio, using the loans' outstanding debt as weights. *Weighted* NPL *vintage* is the weighted average of NPL vintage of the banks lending to a firm, using the lending granted to the firm by each bank as of the end of 2017 as weights. *Risky* is a dummy variable taking value one if the firm's interest coverage ratio is above the median of the distribution as of the end of 2017 and 0 otherwise. See online appendix B for precise definitions of all the variables.

	(1)	(2)	(3)	(4)	(5)
$\log(1+Vintage)$	0.0071	0.0072	$0.0155^{***}$	0.0128***	0.0153***
	(0.0054)	(0.0048)	(0.0033)	(0.0046)	(0.0049)
Policy $\times \log(1 + \text{Vintage})$	0.0172	0.0171	0.0008	$0.0191^{***}$	$0.0166^{***}$
	(0.0124)	(0.0108)	(0.0049)	(0.0055)	(0.0057)
Bank-Time FE	Ν	Υ	Υ	Υ	Y
Firm-Time FE	Ν	Ν	Υ	Υ	Υ
Firm-Bank FE	Ν	Ν	Ν	Υ	Υ
Loan Type FE	Ν	Ν	Ν	Ν	Υ
Observations	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$
R-squared	0.01	0.06	0.60	0.66	0.66

Table 3: Policy and NPL disp	oosals
------------------------------	--------

Notes: The table presents regressions results of a linear probability model at the NPL level, where the dependent variable is a dummy variable that equals 1 if the loan is disposed of in the next quarter as NPL, and 0 otherwise. *Policy* is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. *Vintage* is the number of months the NPL has been classified as such. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are clustered at the bank-quarter level and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.



Figure 2: Policy and NPL disposals: Parallel trends

Notes: This figure uses quarterly data for the period 2017q1 to 2019q3. The dotted line corresponds to the introduction of the ECB NPL Provisioning Expectations (2018q2). The graph plots period-by-period coefficients and 95% confidence intervals that we obtain by replacing in equation 1 the variable *Policy* by a sequence of period (quarter) dummies spanning all periods used in the estimation window. Standard errors are clustered at the bank-quarter level.

	Drop NPLs > 10% decline	Drop low-NPL banks	Drop rural banks & foreign credit institutions	Drop construction & real estate sector
	(1)	(2)	(3)	(4)
$\log(1+\text{vintage})$	$0.0135^{***}$ (0.0049)	$0.0179^{***}$ (0.0053)	$0.0174^{***}$ (0.0057)	$0.0127^{**}$ (0.0052)
Policy $\times \log(1 + \text{vintage})$	0.0206*** (0.0060)	$0.0147^{**}$ (0.0062)	$0.0133^{**}$ (0.0065)	0.0132** (0.0058)
Bank-Time FE	Y	Y	Y	Υ
Firm-Time FE	Y	Υ	Υ	Υ
Firm-Bank FE	Υ	Υ	Υ	Υ
Loan Type FE	Υ	Υ	Υ	Υ
Observations	$1,\!394,\!607$	$1,\!494,\!992$	$1,\!431,\!013$	$1,\!158,\!527$
R-squared	0.69	0.67	0.67	0.62

 Table 4: Effect of policy on NPL disposals - Robustness tests

Notes: This table presents robustness tests for the estimation results of the specification in column (5) of Table 3. The dependent variable is a dummy variable that equals 1 if the loan exists the CIR the next quarter as NPL, and 0 otherwise. *Policy* is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. *Vintage* is the number of months the NPL has been classified as such. Column (1) excludes NPLs whose outstanding debt decreased by more than 10% (at its final quarter in the CIR) at any moment before their disposal. Column (2) presents results with a sample of banks with NPL ratios above 5%. Column (3) presents results where rural banks and foreign credit institutions operating in Spain are excluded from the sample. Column (4) presents results without NPLs belonging to construction or real estate firms. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are clustered at the bank-quarter level and reported in parentheses.

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	(1)	(2)	(3)	(4)	(5)
$Policy \times log(1+Vintage)$	0.0300**	$0.0396^{***}$	0.0302***	$0.0254^{**}$	0.0392***
	(0.0116)	(0.0121)	(0.0117)	(0.0116)	(0.0121)
$Policy \times log(1+Vintage) \times Size$	-0.0065	-0.0109**	-0.0067	-0.0041	-0.0100*
	(0.0053)	(0.0054)	(0.0053)	(0.0051)	(0.0052)
$Policy \times log(1+Vintage) \times ROA$		$0.0373^{***}$			$0.0366^{**}$
		(0.0135)			(0.0152)
$Policy \times log(1+Vintage) \times Capital$			0.0115		-0.0008
			(0.0187)		(0.0201)
$Policy \times log(1+Vintage) \times NPL ratio$				-0.0239*	-0.0022
				(0.0143)	(0.0168)
Bank-Time FE	Y	Υ	Υ	Υ	Y
Firm-Time FE	Υ	Υ	Υ	Υ	Υ
Firm-Bank FE	Υ	Υ	Υ	Υ	Υ
Loan Type FE	Υ	Υ	Υ	Υ	Υ
Observations	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$	$1,\!654,\!107$
R-squared	0.66	0.66	0.66	0.66	0.66

Table 5: Accounting for bank-level heterogeneity in NPL disposal

Notes: The table presents regressions results of a linear probability model at the NPL level, where the dependent variable is a dummy variable that equals 1 if the loan exists the CIR the next quarter as NPL, and 0 otherwise. *Policy* is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. *Vintage* is the number of months the NPL has been classified as such. We interact *Policy* and log(1 + Vintage) with bank characteristics to account for bank heterogeneous effects. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are clustered at the bank-quarter level and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	Log credit	New credit dummy	Termination dummy	Collateralized loan ratio	Long-term loan ratio
	(1)	(2)	(3)	(4)	(5)
Policy×NPL vintage	$-0.0061^{***}$ (0.0019)	0.0002 (0.0011)	$0.0007^{*}$ (0.0004)	$0.0017^{***}$ (0.0003)	$\begin{array}{c} 0.0002 \\ (0.0019) \end{array}$
Bank controls	Υ	Υ	Υ	Υ	Y
Relationship controls	Υ	Υ	Υ	Υ	Υ
Bank-Firm FE	Υ	Υ	Υ	Υ	Υ
Firm-Time FE	Υ	Υ	Υ	Υ	Υ
Bank Type-Time FE	Υ	Υ	Υ	Υ	Υ
Observations	6,776,491	6,776,491	8,284,342	6,776,491	6,776,491
R-squared	0.95	0.65	0.57	0.97	0.86

Table 6: Policy and bank lending standards

This table contains a set of regressions in which the dependent variables are the natural logarithm of outstanding credit from bank b to firm f (column 1), a dummy that equals 1 if bank b extended a new loan to firm f (column 2), a dummy that equals 1 if bank b terminates the lending relationship with firm f (column 3), the ratio of collateralized credit that firm f has with bank b (column 4), and the ratio of bank debt with residual maturity above three years that firm f has with bank b (column 5). Policy is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. NPL vintage is a bank's weighted average vintage of the loan portfolio to non-financial firms as of the end of 2017. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are double clustered at the bank and firm levels and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.



Figure 3: The effect of policy on log credit: Parallel trends

Notes: This figure uses quarterly data for the period 2017q1 to 2019q3. The dotted line corresponds to the introduction of the ECB NPL Provisioning Expectations (2018q2). The graph plots period-by-period coefficients and 95% confidence intervals that we obtain in Equation 2 by replacing the variable *Policy* by a sequence of period (quarter) dummies spanning all periods used in the estimation window. Standard errors are double clustered at the bank and firm level.

	Bank group*quarter dummies	Controls*quarter dummies	Drop low-NPL banks	Drop rural banks & foreign credit institutions	Drop construction & real estate sector
	(1)	(2)	(3)	(4)	(5)
Policy $\times$ NPL vintage	$-0.0059^{***}$ (0.0017)	$-0.0085^{***}$ (0.0020)	-0.0060*** (0.0011)	$-0.0075^{***}$ (0.0019)	$-0.0061^{***}$ (0.0018)
Bank controls	Y	Y	Υ	Υ	Y
Relationship controls	Y	Υ	Υ	Υ	Υ
Bank x Firm FE	Y	Υ	Υ	Υ	Υ
Firm x Quarter FE	Υ	Υ	Υ	Υ	Υ
Bank Type x Quarter FE	Y	Υ	Υ	Υ	Υ
Observations	6,776,491	6,776,491	5,180,135	5,012,521	$5,\!688,\!858$
R-squared	0.95	0.95	0.95	0.95	0.95

Table 7: Policy and bank lending - Robustness tests

Notes: This table presents robustness tests for the estimation results of the specification in column (1) of Table 6. The dependent variable is the logarithm of total credit granted to firm f by bank b. Policy is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. NPL vintage is a bank's weighted average vintage of the loan portfolio to non-financial firms as of the end of 2017. In column (1), we include a bank group dummy interacted with quarter dummies, where the bank group dummy takes the value 1 if the banking group participated in the 2018 EU-wide stress test and 0 otherwise. In column (2), we interact all bank controls with quarter dummies. In column (3), we drop from the sample banks with NPL ratios below 5% (Low-NPL banks). In column (4), we drop from the sample rural banks and foreign credit institutions operating in Spain. Finally, in column (5), we drop firms belonging to the construction or real estate sectors. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are double clustered at the bank and firm levels and reported in parentheses.

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	(1)	(2)	(3)	(4)	(5)	(6)
NPL vintage×Policy	-0.003	0.000	-0.003	0.001	-0.001	0.006
	(0.003)	(0.004)	(0.003)	(0.004)	(0.002)	(0.00)
NPL vintage×Policy×Size	-0.008***	-0.011***	-0.006***	-0.009***	-0.004**	-0.007***
NPL vintage×Policy×ROA	(0.002)	(0.003) $0.007^{**}$ (0.003)	(0.002)	(0.002)	(0.002)	(0.003) $0.009^{**}$ (0.003)
NPL vintage×Policy×Capital		(0.005)	-0.003			-0.006***
			(0.003)			(0.002)
NPL vintage×Policy×NPL ratio				-0.005**		-0.004**
NPL vintage×Policy×NPL coverage				(0.002)	$0.010^{***}$ (0.004)	(0.002) $0.009^{**}$ (0.004)
Bank controls	Y	Y	Y	Y	Y	Υ
Relationship controls	Υ	Υ	Υ	Υ	Υ	Υ
Bank-Firm FE	Υ	Υ	Υ	Υ	Υ	Υ
Firm-Time FE	Υ	Υ	Υ	Υ	Υ	Υ
Bank Type-Time FE	Υ	Υ	Υ	Υ	Υ	Υ
Observations	6,776,491	6,776,491	6,776,491	6,776,491	6,776,491	6,776,491
R-squared	0.95	0.95	0.95	0.95	0.95	0.95

Table 8:	Accounting for	bank-leve	heterogeneity	in bank	lending
----------	----------------	-----------	---------------	---------	---------

The table presents regression results, where the dependent variable is the natural logarithm of outstanding credit from bank b to firm f. Policy is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. NPL vintage is a bank's weighted average vintage of the loan portfolio to nonfinancial firms as of the end of 2017. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are double clustered at the bank and firm levels and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	Growth between the end of 2017 and 2019					
	Bank debt (1)	Employment (2)	Investment (3)	Assets (4)	Sales (5)	
Weighted NPL vintage	-0.0153* (0.0083)	-0.0040*** (0.0007)	-0.0074*** (0.0020)	-0.0021** (0.0010)	$-0.0035^{***}$ (0.0009)	
Province-Industry-Size FE	Υ	Υ	Υ	Υ	Υ	
Firm controls	Y	Υ	Υ	Υ	Υ	
Bank controls	Υ	Υ	Υ	Υ	Υ	
Firm-Bank controls	Υ	Υ	Υ	Υ	Υ	
Observations	113,081	113,081	113,081	113,081	113,081	
R-squared	0.12	0.09	0.09	0.17	0.12	

#### Table 9: Firm-level outcomes

Notes: The table presents regression results, where the dependent variable is the growth between the end of 2017 and 2019 (pre and post policy) of a firm's real variable, presented in columns (1) to (5): bank debt, employment (measured as the number of workers), investment (measured as tangible fixed assets), assets, and sales. Weighted NPL vintage is the weighted average NPL vintage of banks lending to firm f in 2017, taking as weights the amount granted by each bank to firm f. We include province-industry-size fixed effects and controls. Standard errors are clustered at the main bank level and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	Log credit	New credit	Termination	Collateralized loan ratio	Long-term loan ratio
	(1)	(2)	(3)	(4)	(5)
Policy×NPL vintage	-0.0061* (0.0033)	-0.0001 (0.0013)	0.0019 (0.0015)	0.0003 (0.0002)	-0.0028* (0.0014)
$Policy \times NPL$ vintage × Risky	-0.0029** (0.0014)	-0.0023** (0.0011)	-0.0005* (0.0003)	0.0001 (0.0001)	-0.0007* (0.0004)
Policy marginal effect for risky firms	$-0.0090^{***}$ (0.0033)	$-0.0024^{*}$ (0.0013)	$0.0014 \\ (0.0016)$	$0.0004^{*}$ (0.0002)	$-0.0035^{**}$ (0.0017)
Controls	Y	Y	Y	Y	Y
Bank-Firm FE	Υ	Υ	Υ	Υ	Υ
Firm-Time FE	Y	Υ	Υ	Υ	Υ
Bank Type-Time FE	Y	Υ	Υ	Υ	Υ
Observations	$2,\!907,\!652$	2,907,652	3,109,871	$2,\!907,\!652$	2,907,652
R-squared	0.95	0.52	0.54	0.98	0.89

Table 10: Effect on log lending, heterogeneous effects based on firm riskiness

Notes: The table presents regression results, where the dependent variable is the natural logarithm of outstanding credit from bank b to firm f. Policy is a dummy variables that equal 1 for 2018q2 onwards and 0 otherwise. NPL vintage is a bank's weighted average vintage of the loan portfolio to non-financial firms as of the end of 2017. Risky takes the value 1 if the firm's interest coverage ratio is above the median of the distribution as of the end of 2017, and 0 otherwise. The fixed effects that are included in each regression are noted in the lower part of the table. All columns include interactions of NPL vintage, the policy dummy, and firm size dummies. Standard errors are double clustered at the bank and firm levels and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

	Growth between the end of 2017 and 2019					
	Bank debt (1)	Employment (2)	Investment (3)	Assets (4)	$\begin{array}{c} \text{Sales} \\ (5) \end{array}$	
Weighted NPL vintage	-0.0102	-0.0023**	-0.0053**	-0.0015	-0.0033***	
	(0.0088)	(0.0010)	(0.0023)	(0.0012)	(0.0010)	
Risky	$0.0313^{***}$	-0.0116	-0.0011	-0.0207***	0.0011	
	(0.0116)	(0.0093)	(0.0063)	(0.0045)	(0.0073)	
Weighted NPL vintage $\times$ Risky	-0.0110***	-0.0030**	-0.0043**	-0.0010	-0.0005	
	(0.0034)	(0.0014)	(0.0018)	(0.0008)	(0.0015)	
Province x Industry x Size FE	Y	Y	Y	Y	Y	
Firm controls	Υ	Υ	Υ	Υ	Υ	
Bank controls	Υ	Υ	Υ	Υ	Υ	
Firm x Bank controls	Υ	Υ	Υ	Υ	Υ	
Observations	$113,\!078$	113,078	$113,\!078$	$113,\!078$	$113,\!078$	
R-squared	0.1266	0.0917	0.0907	0.1709	0.1192	

Table 11: Firm-level outcomes, heterogeneous effects based on firm riskiness

Notes: The table presents regression results, where the dependent variable is the growth between the end of 2017 and 2019 (pre and post policy) of a firm's real variable, presented in columns (1) to (5): bank debt, employment (measured as the number of workers), investment (measured as tangible fixed assets), assets, and sales. Weighted NPL vintage is the weighted average NPL vintage of banks lending to firm f in 2017, taking as weights the amount granted by each bank to firm f. Moreover, we interact the variable of interest with a firm risk measure. Risky takes the value 1 if the firm's interest coverage ratio is above the median of the distribution as of the end of 2017 and 0 otherwise. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are clustered at the main bank level and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

## **B** Variable definitions

## Loan-level variables (Source: Credit Register, Bank of Spain)

- *Disposal*: A dummy variable that equals 1 if the bank stopped reporting the loan to the CIR, and 0 otherwise.
- *Size*: The natural logarithm of the loan's outstanding debt.
- *Vintage*: The natural logarithm of 1 plus the number of months that the loan is classified as non-performing.

## Firm-bank variables (Source: Credit Register, Bank of Spain)

- log(Credit): The natural logarithm of the granted commitment.
- *NewCredit*: A dummy variable that equals 1 if the bank grants a new loan to the firm.
- *Termination*: A dummy variable that equals 1 if the bank terminates the relationship with the firm.
- Share of secured loans: The amount of collateralized loans divided by total debt.
- *Share of short-term loans*: The amount of loans with residual maturity of less than a year divided by total debt.
- Share of NPLs: The amount of non-performing loans divided by total debt.
- Share of forborne/refinanced loans: The amount of Forborne/refinanced loans divided by total debt.
- Share of loans with bank j: The total amount of loans from bank j divided by firms' total debt.

## Firm-level variables (Source: Spanish Mercantile Register, Bank of Spain)

- *Risky*: A dummy variable that equals 1 if the firm's interest coverage ratio is above the median of the distribution as of the end of 2017, and 0 otherwise.
- *Growth in credit*: The difference in the logarithm of bank committed credit (drawn and undrawn funds) between 2017 and 2019.
- Growth in no. of employers: The difference in the logarithm of the number of workers between 2017 and 2019.
- *Growth in tangible fixed assets*: The difference in the logarithm of tangible fixed assets between 2017 and 2019.
- Growth in assets: The difference in the logarithm of assets between 2017 and 2019.
- Growth in sales: The difference in the logarithm of sales between 2017 and 2019.
- Weighted average NPL vintage: The weighted average NPL vintage of the firm's creditors, using as weights the total amount each bank lends to the firm as of the end of 2017.
- Size: The logarithm of total assets as of the end of 2017.
- Capital ratio: The ratio of book equity to total assets as of the end of 2017.
- Liquidity ratio: The ratio of current assets to total assets as of the end of 2017.
- ROA: The ratio of earnings in 2017 to total assets as of the end of 2017.
- *Tangible assets to total assets*: The ratio of tangible fixed assets to total assets as of the end of 2017.
- Age: The logarithm of 1 plus the age of the firm, measured as the difference between the current year and the date of incorporation.

## Bank-level variables (Source: Supervisory Reports, Bank of Spain)

• *Size*: The logarithm of the bank's total assets.

- *Liquidity*: The ratio of liquid assets (cash and balance with central banks, and loans and advances to governments and credit institutions) held by the bank divided by its total assets.
- NPL ratio: Loans in default as a proportion of the bank's total credit.
- NPL coverage: The ratio of provisions to NPLs.
- *RoA*: Net income divided by assets.
- Capital: Book bank equity divided by total assets.
- *NPL vintage*: A bank's weighted average vintage of the loan portfolio to nonfinancial firms as of the end of 2017.

## C Event study analysis

#### Stock market reactions around NPL guidance announcements

We use conventional event study techniques to test the effect of the announcement of the criteria in the provisioning guidelines on the stock returns of listed Spanish banks. Under the assumption that market anticipations about the ECB's NPL policy before the actual announcement date would have have already been incorporated in the stock prices, one should not expect any significant changes in stock returns around the time of the ECB's policy communication.

We test separately whether the new criteria communicated on March 15th, 2018, and July 11th, 2018, convey new information to the stock markets by focusing on how stock prices of listed Spanish banks reacted to these policy announcements. To do so, we compute the cumulative abnormal returns (CAR) of banks around each policy announcement. Following Correa et al. (2014), we compute abnormal returns,  $r_{b,t}$ , as the difference between realized returns and the expected returns implied by a market model:

$$r_{b,t} = R_{bt} - \hat{\alpha_b} - \hat{\gamma_b} R_{mt}$$

where  $r_{bt}$  is the abnormal return for bank b at time t,  $R_{bt}$  is the realized stock return of bank b, and  $R_{mt}$  is the market return (IBEX35 Bank index). Moreover,  $\hat{\alpha}$  and  $\hat{\gamma}$  are the intercept and slope coefficients of an OLS regression of bank stock returns on a constant and the market return estimated using daily data ranging from 75 days prior to the event to 6 days prior to the event. In other words, we consider an estimation window of [T - 75, T - 6] before each event in our sample. Finally, we compute CARs as the sum of abnormal returns  $(r_{bt})$  over alternative event windows ranging from (-5, +5) (5 days before and after the policy announcements) to [0, 0] (within-day excess return on the day of the event).

The results of this exercise, presented in Table A12, indicate that the Spanish banks experienced significant abnormal losses around the ECB's policy announcements, revealing quantitative aspects of NPL resolutions. Depending on the event window, the estimated CAR for the policy announcement for March 15 and July 11 reach -3.4 percent and -1.79 percent, respectively. These findings support our identifying assumption that the ECB's announcements on March 15, 2018, and July 11, 2018, regarding the quantitative aspects of the NPL provisioning guidelines were highly unanticipated.

Estimation window: $[T - 75, T - 6]$	Cumulative abnormal return (CAR)					
Event window	(-5,5)	(-2,2)	(-1,1)	$(0,\!0)$	(-1,0)	(-1,2)
Policy Announcements on March 15, 2018						
CAR	-0.0340**	$-0.0185^{*}$	$-0.0137^{*}$	-0.0018	-0.0160***	$-0.0174^{**}$
T-stat.	-2.3830	-1.9361	-1.8430	-0.4159	-2.6537	-2.0369
Policy Announcements on July 11, 2018						
CAR	-0.0188	-0.0115	$-0.0179^{**}$	-0.0062	-0.0131**	-0.0166*
T-stat.	-1.2760	-1.1548	-2.3173	-1.3722	-2.0618	-1.8539
Number of banks	8	8	8	8	8	8

Table C.1: Stock market reactions around NPL guidance announcements on March 15, 2018, and July 11, 2018

*Notes:* The table presents the estimation results for cumulative abnormal returns in different event windows. The estimation window is from 75 days before to 6 days before the events. \*, \*\*, and \*\*\* indicate significance at 10 percent, 5 percent, and 1 percent, respectively.

## **D** Additional tables

	Log credit	New credit dummy	Termination dummy	Collateralized loan ratio	Long-term loan ratio
	(1)	(2)	(3)	(4)	(5)
Policy $\times$ NPL vintage	$-0.0062^{***}$ (0.0021)	-0.0001 (0.0012)	$0.0007^{*}$ (0.0004)	$0.0004^{***}$ (0.0001)	-0.0002 (0.0012)
Bank controls	Y	Y	Y	Y	Y
Relationship controls	Y	Y	Υ	Y	Y
Bank-Firm FE	Υ	Υ	Υ	Y	Υ
Firm-Time FE	Υ	Υ	Υ	Υ	Υ
Bank Type-Time FE	Υ	Υ	Υ	Υ	Υ
Observations	2,969,625	2,969,625	3,109,871	2,969,625	2,969,625
R-squared	0.95	0.52	0.54	0.98	0.89

Table D.1: Policy and bank lending standards, sample of firm with balance-sheet variables

This table replicates Table 6 using the sub-sample of firms employed for the firm-level analysis. It contains a set of regressions in which the dependent variables are the natural logarithm of outstanding credit from bank b to firm f (column 1), a dummy that equals 1 if bank b extended a new loan to firm f (column 2), a dummy that equals 1 if bank b extended a new loan to firm f (column 2), a dummy that equals 1 if bank b extended a new loan to firm f (column 2), a dummy that equals 1 if bank b terminates the lending relationship with firm f (column 3), the ratio of collateralized credit that firm f has with bank b (column 4), and the ratio of bank debt with residual maturity above three years that firm f has with bank b (column 5). Policy is a dummy variable that equals 1 for observations in the post-policy period (t > 2018q1) and 0 otherwise. NPL vintage is a bank's weighted average vintage of the loan portfolio to non-financial firms as of the end of 2017. The fixed effects that are included in each regression are noted in the lower part of the table. Standard errors are double clustered at the bank and firm levels and reported in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.10.

#### **BANCO DE ESPAÑA PUBLICATIONS**

#### WORKING PAPERS

- 2310 ANDRÉS ALONSO-ROBISCO, JOSÉ MANUEL CARBÓ and JOSÉ MANUEL MARQUÉS: Machine Learning methods in climate finance: a systematic review.
- 2311 ALESSANDRO PERI, OMAR RACHEDI and IACOPO VAROTTO: The public investment multiplier in a production network.
- 2312 JUAN S. MORA-SANGUINETTI, JAVIER QUINTANA, ISABEL SOLER and ROK SPRUK: Sector-level economic effects of regulatory complexity: evidence from Spain.
- 2313 CORINNA GHIRELLI, ENKELEJDA HAVARI, ELENA MERONI and STEFANO VERZILLO: The long-term causal effects of winning an ERC grant.
- 2314 ALFREDO GARCÍA-HIERNAUX, MARÍA T. GONZÁLEZ-PÉREZ and DAVID E. GUERRERO: How to measure inflation volatility. A note.
- 2315 NICOLÁS ABBATE, INÉS BERNIELL, JOAQUÍN COLEFF, LUIS LAGUINGE, MARGARITA MACHELETT, MARIANA MARCHIONNI, JULIÁN PEDRAZZI and MARÍA FLORENCIA PINTO: Discrimination against gay and transgender people in Latin America: a correspondence study in the rental housing market.
- 2316 SALOMÓN GARCÍA: The amplification effects of adverse selection in mortgage credit suply.
- 2317 METTE EJRNÆS, ESTEBAN GARCÍA-MIRALLES, METTE GØRTZ and PETTER LUNDBORG: When death was postponed: the effect of HIV medication on work, savings and marriage.
- 2318 GABRIEL JIMÉNEZ, LUC LAEVEN, DAVID MARTÍNEZ-MIERA and JOSÉ-LUIS PEYDRÓ: Public guarantees and private banks' incentives: evidence from the COVID-19 crisis.
- 2319 HERVÉ LE BIHAN, DANILO LEIVA-LEÓN and MATÍAS PACCE: Underlying inflation and asymmetric risks.
- 2320 JUAN S. MORA-SANGUINETTI, LAURA HOSPIDO and ANDRÉS ATIENZA-MAESO: The numbers of equality regulation. Quantifying regulatory activity on non-discrimination and its relationship with gender gaps in the labour market.
- 2321 ANDRES ALONSO-ROBISCO and JOSÉ MANUEL CARBÓ: Analysis of CBDC Narrative of Central Banks using Large Language Models.
- 2322 STEFANIA ALBANESI, ANTÓNIO DIAS DA SILVA, JUAN F. JIMENO, ANA LAMO and ALENA WABITSCH: New technologies and jobs in Europe.
- 2323 JOSÉ E. GUTIÉRREZ: Optimal regulation of credit lines.
- 2324 MERCEDES DE LUIS, EMILIO RODRÍGUEZ and DIEGO TORRES: Machine learning applied to active fixed-income portfolio management: a Lasso logit approach.
- 2325 SELVA BAHAR BAZIKI, MARÍA J. NIETO and RIMA TURK-ARISS: Sovereign portfolio composition and bank risk: the case of European banks.
- 2326 ANGEL-IVAN MORENO and TERESA CAMINERO: Assessing the data challenges of climate-related disclosures in european banks. A text mining study.
- 2327 JULIO GÁLVEZ: Household portfolio choices under (non-)linear income risk: an empirical framework.
- 2328 NATASCHA HINTERLANG: Effects of Carbon Pricing in Germany and Spain: An Assessment with EMuSe.
- 2329 RODOLFO CAMPOS, SAMUEL PIENKNAGURA and JACOPO TIMINI: How far has globalization gone? A tale of two regions.
- 2330 NICOLÁS FORTEZA and SANDRA GARCÍA-URIBE: A Score Function to Prioritize Editing in Household Survey Data: A Machine Learning Approach.
- 2331 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Taxing consumption in unequal economies.
- 2332 ESTHER CÁCERES and MATÍAS LAMAS: Dividend Restrictions and Search for Income.
- 2333 MARGARITA MACHELETT: Gender price gaps and competition: Evidence from a correspondence study.
- 2334 ANTON NAKOV and CARLOS THOMAS: Climate-conscious monetary policy.
- 2335 RICARDO BARAHONA, STEFANO CASSELLA and KRISTY A. E. JANSEN: Do teams alleviate or exacerbate the extrapolation bias in the stock market?
- 2336 JUAN S. MORA-SANGUINETTI and ANDRÉS ATIENZA-MAESO: "Green regulation": A quantification of regulations related to renewable energy, sustainable transport, pollution and energy efficiency between 2000 and 2022.
- 2401 LAURA HOSPIDO, NAGORE IRIBERRI and MARGARITA MACHELETT: Gender gaps in financial literacy: a multi-arm RCT to break the response bias in surveys.
- 2402 RUBÉN DOMÍNGUEZ-DÍAZ, SAMUEL HURTADO and CAROLINA MENÉNDEZ: The medium-term effects of investment stimulus.
- 2403 CLODOMIRO FERREIRA, JOSÉ MIGUEL LEIVA, GALO NUÑO, ÁLVARO ORTIZ, TOMASA RODRIGO and SIRENIA VAZQUEZ: The heterogeneous impact of inflation on households' balance sheets.

- 2404 JORGE ABAD, GALO NUÑO and CARLOS THOMAS: CBDC and the operational framework of monetary policy.
- 2405 STÉPHANE BONHOMME and ANGELA DENIS: Estimating individual responses when tomorrow matters.
- 2406 LAURA ÁLVAREZ-ROMÁN, SERGIO MAYORDOMO, CARLES VERGARA-ALERT and XAVIER VIVES: Climate risk, soft information and credit supply.
- 2407 JESÚS FERNÁNDEZ-VILLAVERDE, JOËL MARBET, GALO NUÑO and OMAR RACHEDI: Inequality and the zero lower bound.
- 2408 PABLO BURRIEL, MAR DELGADO-TÉLLEZ, CAMILA FIGUEROA, IVÁN KATARYNIUK and JAVIER J. PÉREZ: Estimating the contribution of macroeconomic factors to sovereign bond spreads in the euro area.
- 2409 LUIS E. ROJAS and DOMINIK THALER: The bright side of the doom loop: banks' sovereign exposure and default incentives.
- 2410 SALOMÓN GARCÍA-VILLEGAS and ENRIC MARTORELL: Climate transition risk and the role of bank capital requirements.
- 2411 MIKEL BEDAYO and JORGE E. GALÁN: The impact of the Countercyclical Capital Buffer on credit: Evidence from its accumulation and release before and during COVID-19.
- 2412 EFFROSYNI ADAMOPOULOU, LUIS DÍEZ-CATALÁN and ERNESTO VILLANUEVA: Staggered contracts and unemployment during recessions.
- 2413 LUIS FÉRNANDEZ LAFUERZA and JORGE E. GALÁN: Should macroprudential policy target corporate lending? Evidence from credit standards and defaults.
- 2414 STÉPHANE BONHOMME and ANGELA DENIS: Estimating heterogeneous effects: applications to labor economics.
- 2415 LUIS GUIROLA, LAURA HOSPIDO and ANDREA WEBER: Family and career: An analysis across Europe and North America.
- 2416 GERALD P. DWYER, BILJANA GILEVSKA, MARÍA J. NIETO and MARGARITA SAMARTÍN: The effects of the ECB's unconventional monetary policies from 2011 to 2018 on banking assets.
- 2417 NICOLÁS FORTEZA, ELVIRA PRADES and MARC ROCA: Analysing the VAT cut pass-through in Spain using webscraped supermarket data and machine learning.
- 2418 JOSÉ-ELÍAS GALLEGOS: HANK beyond FIRE: Amplification, forward guidance, and belief shocks.
- 2419 DANIEL ALONSO: Stabilisation properties of a SURE-like European unemployment insurance.
- 2420 FRANCISCO GONZÁLEZ, JOSÉ E. GUTIÉRREZ and JOSÉ MARÍA SERENA: Shadow seniority? Lending relationships and borrowers' selective default.
- 2421 ROBERTO BLANCO, MIGUEL GARCÍA-POSADA, SERGIO MAYORDOMO and MARÍA RODRÍGUEZ-MORENO: Access to credit and firm survival during a crisis: the case of zero-bank-debt firms.
- 2422 FERNANDO CEREZO, PABLO GIRÓN, MARÍA T. GONZÁLEZ-PÉREZ and ROBERTO PASCUAL: The impact of sovereign debt purchase programmes. A case study: the Spanish-to-Portuguese bond yield spread.
- 2423 EDGAR SILGADO-GÓMEZ: Sovereign uncertainty.
- 2424 CLODOMIRO FERREIRA, JULIO GÁLVEZ and MYROSLAV PIDKUYKO: Housing tenure, consumption and household debt: life-cycle dynamics during a housing bust in Spain.
- 2425 RUBÉN DOMÍNGUEZ-DÍAZ and SAMUEL HURTADO: Green energy transition and vulnerability to external shocks.
- 2426 JOSEP GISBERT and JOSÉ E. GUTIÉRREZ: Bridging the gap? Fintech and financial inclusion.
- 2427 RODOLFO G. CAMPOS, MARIO LARCH, JACOPO TIMINI, ELENA VIDAL and YOTO V. YOTOV: Does the WTO Promote Trade? A Meta-analysis.
- 2428 SONER BASKAYA, JOSÉ E. GUTIÉRREZ, JOSÉ MARÍA SERENA and SERAFEIM TSOUKAS: Bank supervision and non-performing loan cleansing.